Market Power, Bank Funding, and the Transmission of the Monetary Policy to Bank Lending and Profitability *

Ece Özge Emeksiz[†]

University of California Davis

This version: November 2022 Click here for the latest version

Abstract

This paper analyzes how bank market power affects monetary policy transmission to bank funding dynamics, lending, and profitability. First, I document variation in banks' exposure to the monetary policy via spreads on deposits, wholesale funding, and lending, and that bank market power is a strong predictor of the degree of exposure. Specifically, I show that after an increase in the policy rate, banks with higher market power adjust their deposit and loan rates relatively less, offsetting the fall in their deposit inflows through cheaper access to wholesale funding. This dampens the effect of contractionary monetary policy on their lending and profitability. That is, I present unified evidence on monetary pass-through to the U.S. commercial banks by comprehensively studying the interactions among the deposit, wholesale funding, and credit markets which is missing in the literature. Third, I show that bank market power has implications for monetary policy transmission to the real economy through its impact on bank-level lending. In particular, aggregate lending and employment decrease less in areas served by banks with higher market power following monetary contraction. Finally, I rationalize my empirical findings by building a theoretical model with monopolistic competition where market power generates imperfect passthrough of monetary policy.

^{*}I am grateful to my advisors Alan Taylor, James Cloyne, and Katheryn Russ as well as Oscar Jordá, Sanjay Singh and Nicolas Caramp for invaluable support and feedback throughout this project. I would also like to thank seminar participants at the University of California, Davis for helpful comments and discussions. All errors are my own.

[†]Department of Economics, University of California, Davis (eoemeksiz@ucdavis.edu)

1 Introduction

Striking economic phenomenons of the recent decade in the U.S. banking industry have been the decrease in the banking competition and the growth of money market funds which are the leading players in the U.S. wholesale funding markets. In particular, the market share of the top five banks has increased significantly from less than 25% in the 1990s to over 45% in the last few years. Moreover, the total assets of money market funds nearly tripled during the same time, as shown in Figure 1. However, the current literature studied the effect of market power in isolation, focusing only on its impact on deposit and credit markets separately, and the role of market power on banks' cost of accessing wholesale funding and wholesale funding reliance is overlooked. Thus, a full understanding of how market power mediates monetary policy transmission to bank lending and profitability has been missing.

This paper examines how bank market power affects monetary policy transmission to banks' funding dynamics, lending, and profitability. First, I explore how monetary policy alters banks' funding composition between deposits and wholesale funding and whether bank market power is associated with variation in banks' exposure to monetary policy.¹ I further investigate how the change in banks' funding dynamics, in other words, change in their funding spreads and the funding composition affects monetary pass-through to bank lending spreads, interest margins, and bank profitability depending on the degree of banks' market power. In doing so, I provide novel evidence on the impact of bank market power on the transmission of monetary policy to the U.S. banking sector by jointly studying the deposit, wholesale funding, and credit markets which is critical to reaching accurate insights on the monetary policy transmission mechanism to banks' balance sheets as these markets are highly interconnected. Moreover, I show that bank market power affects the transmission of monetary policy to the macroeconomic outcomes through its impact on bank-level lending. Thus, my results have important implications for how we should think about monetary policy transmission mechanism to the real economy.

The first main contribution of this paper is documenting the dispersion in the response of banks' deposit and wholesale funding spreads to monetary policy depending on the banks' market power.² I show that monetary policy changes the fund-

¹Wholesale funding refers to banks' funding sources other than retail deposits.

²Deposit spread is the difference between federal funds and deposit rates. Similarly, wholesale funding spread is defined as the difference between federal funds and wholesale funding rates. Widening of these spreads implies a partial pass-through of the policy rate to deposit and wholesale funding rates.

ing composition of the banks by creating a wedge in deposit and wholesale funding spreads among banks with different levels of market power. In particular, after monetary policy tightening, deposit and wholesale funding spreads increase more for banks with higher market power operating in highly concentrated local banking markets. In other words, these banks do not fully transmit the increase in interest to their depositors and access wholesale funding markets at a comparatively lower cost. As deposit spreads widen, households switch to products that offer higher yields, with deposit inflows decreasing slightly more for banks with market power as a result. At the same time, these banks considerably increase their wholesale funding dependence and compensate for the decrease in their deposit inflows through wholesale financing. Hence, I show that the funding composition of banks changes substantially in response to monetary policy and banks' market power is a good predictor of this compositional change.

My second main contribution is documenting that the change in bank funding dynamics affects monetary policy transmission to bank *lending* spreads and thus to bank lending.³ Specifically, I show that banks with higher market power increase their loan spreads less following a monetary contraction. Two forces drive this result. First, market power enables banks to increase their deposit spreads more following a contractionary monetary policy. Second, banks with higher market power are able to access wholesale funding at a lower cost. The ability to keep deposit rates relatively low while leaning more on wholesale funding (where they get more favorable terms than other banks) enables banks with the higher market power to raise their interest on loans less as the federal funds rate rises. Thus, these banks' lending decreases significantly less compared to banks with lower market power.⁴ Taken together, my results show that market power dampens the impact of monetary policy on lending by altering the funding dynamics of the banks.

The third contribution of this paper is to provide a unified framework to analyze the effect of bank market power on the transmission of monetary policy to banks' deposit, wholesale funding, and credit decisions which is essential to achieve a complete understanding of monetary policy transmission mechanism to bank lending due to interdependence of these markets. Prior to my work, little was known about the

³Lending spread is defined as the difference between the loan rate and the Treasury yield with a similar maturity.

⁴Figure 2 illustrates this mechanism in detail where Figure 2a plots the average effect of monetary policy on bank funding dynamics and lending and Figure 2b shows the impact of monetary policy on the banks with higher market power.

simultaneous effect of market power on funding and lending markets in the face of a change in the policy rate. I bridge the deposit and credit market power channels that study the impact of market power on either deposit or loan spreads separately. Additionally, I reveal the importance of the wholesale funding channel for the monetary policy transmission mechanism by documenting the heterogeneous response of banks' wholesale funding spreads to monetary policy, which has not been established previously. In particular, I show that bank market power lowers the pass-through of monetary policy into wholesale funding, deposit, and loan rates. Moreover, I find that the lending and net interest margins of banks with higher market power decrease less following a contractionary monetary policy, leading to a relative increase in the profitability of banks with higher market power.

The fourth contribution of this paper is to show that bank market power affects the monetary policy transmission mechanism to the macroeconomy. Previous literature argues that after an increase in the policy rate, deposits flow out of the banking system, and banks with deposit market power shrink their lending. This amplifies the impact of contractionary monetary policy on real economic outcomes (Drechsler, Savov, and Schnabl, 2017). However, they overlooked the effect of market power on banks' wholesale funding and lending spreads. I complete their story by comprehensively analyzing the impact of bank market power on U.S. banking markets and substantially expanding their sample. I show that lending of banks with higher market power decreases significantly less following a contractionary monetary policy as these banks ubstitute wholesale funding for deposit outflows, and pass-through to loan spreads are substantially lower for these banks. As a result, aggregate lending decreases less in areas served by banks with higher market power, and unemployment increases less in these regions following a contractionary monetary policy.

Lastly, I contribute to the literature by building a theoretical model of monopolistic competition both in deposit and credit markets consistent with the micro-foundations presented in Ulate (2021). I use the model to rationalize my empirical findings and explain the underlying mechanism. In the model, banks with higher market power access wholesale funding markets at a lower cost, which generates imperfect pass-through of monetary policy to their deposit, loan rates, and lending confirming my empirical findings. I cross-validate the model's predictions using data from U.S. Call Reports and show that the model performs well in matching the data.

There are a set of empirical challenges in identifying the impact of market power

on the transmission of monetary policy to bank-level and macroeconomic outcomes. One significant identification challenge I face is the potential endogeneity of monetary policy. To address this concern, I use high-frequency monetary shocks from Bauer and Swanson (2022) as an instrument for the one-year Treasury rate in my analysis. These monetary policy shocks satisfy both instrument validity and exogeneity conditions as they are correlated with movements in the one-year Treasury rate yet uncorrelated with all other shocks.

Secondly, banks may have different lending opportunities, and banks' funding decisions might be responding to contemporaneous changes in bank-specific lending opportunities rather than directly to monetary policy. For example, if banks' lending opportunities reduce following the tightening of the monetary policy, banks issue fewer loans, and thus, their reliance on deposits decreases independent of banks' market power (Drechsler, Savov, and Schnabl, 2017). As a result, banks' retail rates might be affected by the change in the bank-specific loan demand rather than the banks' market power. I tackle this issue by exploiting the geographic variation in the concentration of local banking markets and show that my results go through even I compare the funding and lending spreads of the same bank located in areas with different levels of concentration. I implement this branch-level analysis using branch-level deposit and loan rate information on U.S. banks from Ratewatch. I construct my branch-level market power measure by relying on the Herfindahl index (HHI), calculated by summing up the squared asset-market shares of all depository institutions that operate branches in a given county in a given year. As decisions related to wholesale funding are made at the bank rather than the branch level, I conduct my analysis on wholesale financing at the bank level and provide a complete picture of the mechanism using bank-level data from Call Reports.

Another critical identification challenge I face is the different local lending opportunities that banks may face, irrespective of their market power. That is, monetary policy may distinctly affect local lending opportunities in different regions where banks operate. To address this issue, I use Small Business Lending data from the Federal Financial Institutions Examination Council (FFIEC), which reports a bank's lending in a given county for a particular year. To separate the effects of higher rates from the underlying macro environment, I add county-time fixed effects to my analysis which absorbs the average impact of macroeconomic variables on demand for loans. That is, I ensure that my results are not driven by the differences in local lending opportunities. I then aggregate my lending data to the county level and establish a link between bank lending and county-level real economic outcomes. The county-level analysis reveals that bank market power substantially impacts monetary policy transmission through bank balance sheets.

The rest of this paper is organized as follows: Section 2 discusses the related literature, and Section 3 documents the relationship between the federal funds rate and aggregate deposit and wholesale funding flows, providing motivating evidence for my analysis. Section 4 describes the data and provides summary statistics. Section 5 explains the identification strategy and provides the empirical results. Section 6 presents the model. Section 7 provides robustness checks, and Section 8 concludes.

2 Literature Review

This paper contributes four strands of literature. First, it contributes to the literature on the bank lending channel of monetary policy (Bernanke, 1983; Bernanke and Blinder, 1988; Kashyap and Stein, 1994; Kashyap and Stein, 2000) which explores how bank lending responds to change in the monetary policy. In these papers, the underlying channel for the transmission of monetary policy to bank lending operates through bank reserves. On the other hand, the primary mechanism that drives the bank lending dynamics to monetary policy based on banks' market power. Specifically, the monetary policy creates a dispersion in the wholesale funding and deposit spreads of banks with different degrees of market power, which in turn changes the funding composition of the banks and influences the monetary policy pass-through to bank loan rates and bank lending outcomes.

Second, my paper bridges the literature that examines the role of bank market power on the transmission of monetary policy to the bank deposit and lending spreads. These papers have emphasized the importance of bank competition for the monetary policy pass-through to household deposits and mortgage rates (Scharfstein and Sunderam, 2016; Drechsler, Savov, and Schnabl, 2017; Balloch and Koby, 2019; Wang et al., 2022). Yet, they examine the impact of market power either on deposit or loan spreads and overlook its effect on the cost of accessing wholesale funding and banks' funding composition. Specifically, Drechsler, Savov, and Schnabl (2017) only focus on the impact of market power on deposit spreads centering on the pre-ZLB period. On the other hand, Scharfstein and Sunderam (2016) solely investigate the role of market power on mortgage rates concentrating on the low-interest rate environment. Building on these studies, my paper explores the impact of market power on monetary policy passthrough to bank funding and lending spreads jointly. It also proposes a new channel of monetary policy transmission mechanism through wholesale funding.

Third, my paper connects to literature that studies the impact of monetary policy on wholesale funding markets. Particularly, Xiao (2020) and Choi and Choi (2021) show that monetary tightening reduces the supply of retail deposits and expands funding creation in the money markets. I contribute to this literature by studying the impact of bank market power on banks' cost of accessing wholesale funding and banks' funding composition. I document that banks with higher market power change their funding dynamics to increase their funding profits which in turn allows them to charge lower loan markups and smooth the negative effect of contractionary monetary policy on their lending. This is a new dimension that these studies have not addressed, and my results provide significant insights into the impact of bank market power for the monetary policy transmission mechanism.

Finally, my paper connects to the literature that examines the effect of monetary policy on credit costs and real economic outcomes. Gertler and Karadi (2015) document that credit costs increase after a contractionary monetary policy shock due to the rise in the risk premia leading to a contraction in output. Eggertsson et al. (2019), Ulate (2021), Brunnermeier and Koby (2018) and Heider, Saidi, and Schepens (2019) explore the impact of low and negative interest rate environments on loan spreads and bank profitability. They show that the negative rate environment leads to a decline in banks' net worth, which can deteriorate output growth. I contribute to this literature by showing that bank market power alleviates the adverse effects of monetary policy both on bank funding and credit costs. Hence, the lending of banks with higher market power decreases relatively less after an increase in the policy rate. This, in turn, mitigates the negative impact of interest rate hikes on regional macroeconomic outcomes.

3 Motivating Evidence

In this section, I use aggregate bank-level data and document that monetary policy creates a wedge between the banks' funding spreads and the policy rate, leading to a change in the funding composition of the banking sector. Moreover, I illustrate the relationship between the policy rate and the bank profitability.

Figure 3a plots the average deposit and wholesale funding rate for the U.S. commercial banks against the federal funds rate over time using Call Reports data.⁵ The figure reveals that both the deposit and wholesale funding rates rise less than one-forone with the fed funds rate, generating a spread between the banks' funding rates and the policy rate.⁶ As seen from the figure, both the deposit spread ($i^{FFR} - i^D$) and the wholesale funding spread ($i^{FFR} - i^{WF}$) are cyclical, increasing as the federal funds rate rises. Mainly, the tightening of the monetary policy increases the demand for interestbearing deposits and wholesale funding products with respect to the money, enabling banks to restrict the pass-through of the higher policy rate to the cost of their funding.⁷

Figure 3c plots the average deposit rate by different deposit products using branchlevel deposit data from Ratewatch. On average, time deposits offer higher rates than saving deposits as these products are less liquid than saving deposits.⁸ Similarly, the pass-through to the wholesale funding rate is higher compared to the pass-through to the average deposit rate due to the illiquidity risk it bears.⁹

As the federal funds rate rises during monetary policy tightening cycles, the opportunity cost of holding deposits also rises. Consequently, depositors switch to relatively less liquid products that offer higher yields, such as bonds and money market products. This result can be seen by the negative relationship between the banks' core deposits and the federal funds rate presented in Figure 4a. Specifically, the core deposits of the commercial banking system shrink substantially when the federal funds rate increases.¹⁰ In return, banks seek to compensate for the decrease in their deposits by turning to wholesale funding markets, both of which contribute to the expansion of the shares of money market mutual funds, as shown in Figure 4a. Figure 4b plots the time series of wholesale funding to deposit ratio against the federal funds rate. Importantly, there is a positive relationship between the federal funds rate and banks' wholesale funding reliance, indicating that banks substitute wholesale funding for de-

⁵I calculate the deposit rate as interest income on domestic deposits divided by domestic deposits and then annualized.

⁶Drechsler, Savov, and Schnabl (2017) also reported the spread between the federal funds rate and deposits.

⁷Xiao (2020) also reported the spread between the federal funds rate and money market products.

⁸Particularly, time deposits are locked in for a term, whereas checking and saving deposits can be withdrawn immediately, generating a liquidity premium between these products (Drechsler, Savov, and Schnabl, 2017).

⁹Wholesale funding is generally an uninsured form of borrowing, whereas retail deposits are fully insured. For instance, foreign deposits, which are a sizeable part of wholesale financing, are not guaranteed by FDIC.

¹⁰Banks' core retail deposits are calculated as the sum of the transaction and saving deposits.

posit outflows during periods of high-interest rates. Overall, the aggregate level analysis suggests that monetary policy changes the funding composition of the banking sector by generating a dispersion between the policy rate and bank funding rates.

Finally, Figure 3d plots the banks' net interest margins (NIM) and return on assets (ROA) against the federal funds rate and the 10-year Treasury yield. As seen from the figure, the term premium, in other words, the difference between the 10-year Treasury yield and federal funds rate, decreases when the federal funds rate increases. During the same time, NIMs and ROA also slightly decrease. As banks borrow short and lend long, an unexpected increase in the short rate increases banks' interest expense relative to their interest income, reducing their net interest margins. Taken together, the analysis suggests that the banks' profitability is significantly affected by the change in the policy rate.

4 Data and The Summary Statistics

This section describes the data and provides summary statistics relevant to my analysis.

4.1 Retail rates

I use weekly data on loan and deposit rates collected across U.S. bank branches by Ratewatch. Ratewatch provides high-quality information on weekly deposit and loan rates of various deposit and loan products at the branch level. The data spans from January 2000 to December 2019 and can be merged with other data sets using an FDIC branch identifier.¹¹ Using Ratewatch data for my analysis is advantageous for a couple of reasons. First, it has the most extensive product coverage among the available datasets. Specifically, it covers rates on adjustable and fixed-rate mortgages with different maturities, home equity lines of credit (HELOCs), and home equity loans with different LTVs, automobile loans, and personal loans for a specific constant loan volume. On the deposit side, it provides data on savings, time, and checking deposits with various account sizes, such as money market deposit accounts with an account size of \$10,000, \$25,000, \$75,000, and 6-month, 12-month, 24-month certificates of deposit with an account size of \$10,000, \$25,000, etc.

¹¹Drechsler, Savov, and Schnabl (2017) report that deposit data is available starting from 1997. However, Ratewatch provided me data beginning in 2000.

4.2 Small business lending

I collect county-level data on bank lending to small businesses from the Federal Financial Institutions Examination Council (FFIEC). The data covers small business lending by bank and county annually from 1997 to 2019.¹² In particular, it provides information about loan origination to U.S. small businesses (loans smaller than \$1 million in size) at the county level by banks with assets roughly exceeding \$1.25 billion.¹³ I include all bank-county observations with at least \$100,000 of new lending.¹⁴ Small business lending data is particularly convenient for my analysis as small businesses have a strong dependency on local banks (Bord, Ivashina, and Taliaferro, 2015) and an illiquid form of lending. Moreover, it is well suited to investigate the impact of lending on real economic outcomes as small businesses represent more than 90% of all business establishments and around 50% of U.S. GDP. ¹⁵

4.3 Monetary Policy Shocks

I use the Bauer and Swanson (2022) monetary policy shock series, graciously shared by the authors. These shocks are obtained by taking the first principal component of the changes in the first four quarterly Eurodollar futures contracts, ED1–ED4, around the FOMC announcements. Hence, these shocks also capture a forward guidance component, as argued in Gürkaynak, Sack, and Swanson (2005).¹⁶ These series are summed to a quarterly or an annual frequency, and they span from 1988 to 2019.¹⁷ I instrument the changes in the one-year Treasury rate, my policy measure, with the Bauer and Swanson (2022) shocks. I re-scale it so that its effect on the one-year nominal Treasury yield equals one. I further checked the robustness of my results using the alternative monetary policy shocks obtained by orthogonalizing the Bauer and Swanson (2022) shocks with respect to macroeconomic and financial data that pre-date the announcement.¹⁸ There are many alternative approaches to identifying monetary policy

¹²I exclude 2008 from my analysis to ensure that the financial crises do not drive my results, small business lending dropped significantly during the financial crises, as shown in Figure 5b.

¹³It excludes very small banks.

¹⁴My results are robust to using all bank-county observations in the sample.

¹⁵Small Business Administration January 2012. "Small Firms, Employment and Federal Policy," Congressional Budget Office, March 2012.

¹⁶Gürkaynak, Sack, and Swanson (2007) show that Eurodollar futures are the best predictor of future values of the federal funds rate at horizons beyond six months and are virtually as good as federal funds futures at horizons less than six months.

¹⁷County-level data is available only at an annual frequency whereas bank-level data is mostly available at quarterly frequency.

¹⁸Bauer and Swanson (2022) show that shocks obtained through both methods yield similar results on financial variables, whereas the orthogonalized shocks improve the results on macroeconomic vari-

shocks in the monetary policy literature. One of the many novel approaches is using residuals from a regression of the federal funds rate on lagged values and the Federal Reserve's information set using Greenbook forecasts as in Romer et al. (1990). Another approach is to identify the shocks in an SVAR and SVAR-IV as in Gertler and Karadi (2015), Miranda-Agrippino and Ricco (2021). However, many of these measures are not fully available for my period of study.

4.4 Federal funds rate and Treasury yields

In my analysis, I choose to use the one-year Treasury yield as the policy indicator since the average maturity of loans is higher than one year, whereas the average maturity of deposits is close to one year in the data. I instrument the one-year Treasury yield with Bauer and Swanson (2022) shocks to eliminate any concerns about the exogeneity of the policy instrument.¹⁹ I collect the quarterly and yearly government Treasury bills and Federal fund rates from the FED H.15 series. In addition, I obtain data on U.S. commercial paper and 30-year mortgage rates from FRED.

4.5 Macroeconomic Data

The data on national level GDP, inflation and unemployment are from FRED. Similarly, data on U.S. commercial paper spread and 30-year mortgage rates also obtained from FRED. ²⁰

4.6 County data

County-level data on GDP is collected from the U.S. Bureau of Economic Analysis at an annual frequency; it covers January 2001 to December 2019. County-level data on unemployment and wages are obtained from the Bureau of Labor Statistics. It covers 1997 January to 2019 December at an annual frequency.

ables.

¹⁹High-frequency monetary policy shocks are exogenous with respect to all macroeconomic variables that are publicly known prior to the FOMC announcement itself, making them a valid instrument.

²⁰Commercial paper spread refers to difference between the 3-month commercial paper and the federal funds rate.

4.7 Deposit holdings

The data on branch-level deposits is from the Federal Deposit Insurance Corporation (FDIC). The data covers the universe of U.S. bank branches annually from January 1994 to December 2019. The data set also has branch characteristics such as the parent bank, address, and geographic coordinates. I use the unique FDIC branch identifier to match it with other data sets.

4.8 Bank data

Quarterly aggregate bank-level data is obtained from the U.S. Call Reports provided by the Wharton Research data services (WRDS). I use data from January 1997 to December 2019.²¹ The data contains novel information on income statements and balance sheets of all U.S. commercial banks. I match the bank-level Call Reports to the branchlevel Ratewatch data using the FDIC bank identifier.

I focus on the period between 2000 to 2019 in my analysis as branch-level retail rate data is available starting from 2000. I exclude the period of financial crises from my analysis as banks' funding and lending decisions may change for other reasons unrelated to monetary policy during the extreme time of financial distress.

4.9 Summary Statistics

This section provides summary statics at the county, branch, county-bank, and bank levels.

In my empirical approach, I use the Herfindahl-Hirschman Index (HHI) as a proxy for the local banking market concentration. The HHI is calculated by summing up the squared asset shares of all banks that operate branches in a given county in a given year and then averaged over time.²²

$$HHI_{ct} = \sum_{j \in c} \left(\frac{assets_{jct}}{\sum_{j \in c} assets_{jct}} \right)^2$$
(1)

²¹I completed the missing series from data provided by Federal Financial Institutions Examination Council (FFIEC) 's website.

²²As branch-level asset information is unavailable, I use banks' deposit-to-assets ratio from the Call Reports. I combine it with the branch-level deposit data from the Federal Deposit Insurance Corporation (FDIC) and compute the branch-level assets for each year.

I then assign to each bank branch in my data the HHI of the county in which it is located and refer to it as the HHI of the branch. For instance, I calculate the HHI of Miami as 0.11. Then I assign both Bank of America's Miami branch and Citibank's Miami branch an HHI of 0.11.²³

Figure 6a maps the average HHI across the U.S. A lower number indicates a lower concentration level, hence a higher level of competition. There is a significant cross-sectional variation across counties, from a minimum HHI of 0.06 to a maximum of 1. Similarly, Figure 6b maps the same measure using the deposit share of the branches. Notably, both measures are highly correlated and indicate a similar dispersion in concentration among counties. On average, highly concentrated counties are smaller with a lower GDP and income. Moreover, the unemployment rate is slightly higher in these counties.

The loan spread, $(i^L - i^{UST})$, is computed quarterly as the difference between the loan rate paid on a given type of loan and the interest rate on Treasury yield with the respective maturity.²⁴ Appendix A provide details on the construction of bank-level variables. Table 2 Panel B to D report the change in the loan spread by various loan products using branch-level loan data. There is a substantial variation among the bank branches, indicated by high standard deviations.

The deposit spread, $(i^{FFR} - i^D)$, is computed quarterly as the difference between the fed funds rate and the rate paid on a given type of deposit.²⁵ Table 2 Panel A reports the change in the average deposit spread using branch-level deposit data from Ratewatch. The deposit spread varies between 0.01 to -0.03 with a standard deviation around 0.30 to 0.37, indicating a sizeable variation across branches.

Table 3 Panel A provides summary statistics at the bank level. For my bank-level analysis, I compute a bank-level measure of concentration, Bank-HHI, defined as the weighted average of Branch-HHI (HHI_c) across all branches, using branch-level assets as weights.²⁶ The Table indicates greater variation in the wholesale funding spread, ($i^{FFR} - i^{WF}$), computed as the difference between the fed funds rate and the average

²⁶BankHHI_{jt} =
$$\sum_{i \in j} \left(\frac{assets_{it-1}}{\sum_{i \in j} assets_{it-1}} \times HHI_{ct} \right)$$
 as in Equation (3).

²³I use the asset market share of the branches as the main focus of the paper is capturing the market power on all markets. However, my results are robust to using deposit market shares as in Drechsler, Savov, and Schnabl (2017).

²⁴The loan rate is calculated by dividing total interest income on loans by the volume of loans and annualized.

²⁵The deposit rate is calculated by dividing total interest expense on domestic deposits by the volume of domestic deposits and annualized.

rate paid on wholesale funding, is greater than in the deposit and loan spreads across the banks. Figure 5a plots the composition of wholesale funding for the U.S. commercial banks. Foreign Deposits, other borrowed money, brokered deposits, and repos have the highest share in banks' wholesale funding, respectively. In particular, they constitute more than 75% percent of banks' aggregate wholesale financing. Other components of wholesale funding include trading liabilities and subordinated debt, where subordinated debt comprises only a small portion of wholesale funding.

Table 3 Panel B provides summary statistics on small business lending, reported at the bank-county level at an annual frequency. Figure 5b plots the time-series of bank lending to small businesses both for the total volume and the number of loans. Notably, both series have declined substantially during the financial crises and have barely picked up to pre-crisis levels as of 2010. Table 3 Panel C provides information on annual county-level lending, GDP, unemployment, and wages. The average countylevel GDP is around \$5 million, whereas average county-level wages are around \$2 million. The unemployment rate is around %6 with a standard deviation around %2.

The following section provides the empirical framework that explores the role of bank market power on the transmission of monetary policy to bank funding and lending rates, profitability, and lending. Moreover, it shows that monetary policy pass to real economic outcomes through banks' balance sheets.

5 Empirical Analysis

My theory suggests that banks that operate in highly concentrated local banking markets, namely banks with higher market power, increase their interest rates on deposits and loan rates less after a monetary contraction. Moreover, they compensate for the decrease in deposits by increasing their reliance on wholesale funding as they pay relatively less for such funds. Consequently, lending and profitability of banks with higher market power decrease less. Moreover, real economic outcomes are less adversely affected by monetary policy in regions served by banks with higher market power.

Testing my hypothesis is particularly challenging as one cannot establish a direct causal effect of monetary policy on bank retail rates due to the potential for omitted variables. One of the most prominent omitted variables is the change in bank lending opportunities. If banks' lending opportunities changes as the Fed raises rates, this may affect banks' funding and lending decisions independent of banks' market power. In particular, banks' lending may shrink due to the adverse impact of monetary policy on their bank-specific loan demand, leading to a decrease in their funding needs (Drechsler, Savov, and Schnabl, 2017). Thus, the change in lending opportunities may affect the funding and lending rates of the banks, irrespective of their market power. To obtain variation in concentration independent of bank-specific lending opportunities, I compare the deposit and lending rates across branches of the same bank located in areas with different concentration levels.²⁷ Comparing across branches of the same bank enables me to control the bank-specific lending opportunities and assess the effect of concentration on the responsiveness of bank retail rates to monetary policy.

As banks conduct decisions related to wholesale financing at the bank level and allocate their funds internally across their branches if needed, I turn into the bank-level Call Reports data. That is, I test the mechanism that generates an imperfect monetary pass-through to banks' lending rates and dampens the impact of contractionary monetary policy on banks' lending outcomes at the bank level. This bank-level estimation strategy is especially significant to emphasize the importance of the wholesale funding channel and give a complete view of the effect of market power on the transmission of monetary policy to bank balance sheets.

To establish a causal relationship between bank lending outcomes and the real economy, I further use bank-county level small business lending data that reports small business lending by bank and county for a given year. Small business lending is a significant form of borrowing for local businesses, allowing me to show a connection between bank-level lending and regional macroeconomic outcomes. First, I examine the effect of bank market power on banks' small business lending through my bank-county estimation, which enables me to link bank-level lending outcomes to county-level lending outcomes. Then, I assess the implications of the change in county-level lending on real economic outcomes using county-level unemployment data. This county-level analysis reveals that monetary policy transmits to macroeconomic outcomes through the bank lending channel.

Sections 5.1, 5.2, 5.3 and 5.4 present these estimation strategies and report the results.

²⁷This with-in bank estimation strategy comes from Drechsler, Savov, and Schnabl (2017).

5.1 Branch-level Estimation

In order to assess the impact of market power on the transmission of monetary policy to bank funding and lending spreads, I exploit the geographic variation in market power induced by the differences in the concentration of the local banking markets by using branch-level deposit and loan data from Ratewatch. As discussed earlier, the main challenge to identification is isolating the effect of bank market power independent of bank-specific lending opportunities. For instance, banks' lending might decrease because of the negative impact of monetary policy on their bank-specific loan demand, generating a decline in banks' need for deposits and wholesale funding. Thus, the funding and lending rates of the banks might be affected by the change in lending opportunities rather than bank market power. I tackle this issue by comparing the funding and lending spreads of the same bank located in counties with different degrees of concentration. Furthermore, I add state-time fixed effects to my analysis to ensure that these banks are subject to similar local banking market conditions, e.g., similar local loan demand. Finally, to eliminate the concerns regarding the endogeneity of the policy indicator, I instrument the change in the one-year Treasury yield with the plausibly exogenous monetary shocks of Bauer and Swanson (2022).

Equation (2) presents the baseline regression, allowing me to capture the average effect of monetary policy on bank lending and funding spreads in addition to the impact of banking market concentration.

$$\Delta y_{it} = \delta_i + \gamma_c + \lambda_s + \beta_1 \Delta R_t + \beta_2 \Delta R_t \times HHI_c + \Omega'(L)Z_t + \epsilon_{it} \tag{2}$$

 Δy_{it} is either the quarterly change in the deposit spread or the change in the loan spread of branch *i* of bank *j* operating in county *c* from *t*-1 to *t*. ΔR_t is the quarterly change in the one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. *HHI_c* is the concentration of the county where branch *i* of bank *j* is located. It is calculated by summing up the squared asset-market shares of all banks that operate in a given county in a given year and then averaged across years, as shown in Equation (1). δ_i , γ_c , and λ_s are branch, county, and state fixed effects, respectively. ²⁸

Branch fixed effects control for branch-specific characteristics. County fixed effects control for county-specific factors such as county-wide economic trends.²⁹ Similarly,

²⁸The interaction term is also instrumented by the Bauer and Swanson (2022) shocks in this and all the upcoming specifications.

²⁹County and branch fixed effects are highly collinear as only a really small fraction of branches

state-fixed effects control state-specific factors such as time-in-varying banking market conditions. I also add a county fixed effects interacted with a dummy variable for the zero-lower bound period to control for differences that may stem from the zero lower bound period. Z_t includes additional control variables, which include GDP growth, unemployment rate, and inflation.³⁰ The macroeconomic control variables were added to isolate the role of interest rates from cyclical conditions. Finally, I cluster the standard errors at the county level to control for correlation within counties. I compute the interaction terms and control variables relative to their means in this and all my upcoming specifications. The reason is that, by demeaning the variables in this way, the intercept term β_1 has the natural and desirable interpretation as the average conditional path when all controls are at their mean levels.

I further add bank-time fixed effects, which absorb all time-varying differences between banks and allow me to compare the branches of the same bank located in areas with different concentration levels following Drechsler, Savov, and Schnabl (2017). That is, I control for changes in bank-specific lending opportunities that the change in the monetary policy might cause. I also add state-time fixed effects to control statelevel changes in local deposit and lending markets. Finally, I run several specifications with different combinations of fixed effects to gauge their impact and check the robustness of my results.³¹

Table 4 shows that, on average, banks increase their deposit spreads, $(i^{FFR} - i^D)$ after an increase in the policy rate, and deposit spreads widen more for branches in more concentrated areas. That is, the pass-through of monetary policy to deposit rates decreases with bank concentration. Column (1) focuses on all bank branches in my sample and examines the spread dispersion across the branches of the different banks, whereas columns (2) to (5) focus on the banks that operate in at least two counties to make sure that the county-specific community banks do not drive my results. In particular, Columns (3), (4), and (5) add state-time fixed effects, bank-time fixed effects, and both state-time and bank-time fixed effects, respectively.

Table 4 Panel A reports the results for 12 months of certificates of deposits, a common type of small-time deposit. Column (1) documents that deposit spreads increase around 34 bps after a 100 bps raise in the one-year Treasury yield. That is, the av-

change counties.

³⁰Results are robust, adding the change in the commercial paper spread and 30-year mortgage spread.

³¹Adding time fixed effects absorbs the β_1 as the change in the interest rate is the same for all branches at a particular point in time. On the other hand, it allows me to control for underlying observable and unobservable systematic differences between observed time units.

erage deposit rate on 12-months CDs increases by 66 bps after a 100 bps increase in the policy rate. The interaction term on the change in the policy rate and the concentration index indicates that deposit spreads increase 13 bps more for branches that operate in more concentrated counties. In other words, deposit rates increases around 3.9 bps (13*0.30=3.9) less in bank branches that serve in counties with a 0.30 higher Herfindahl-Hirschman Index. Where 0.30 refers approximately the 75th percentile of the HHI distribution. Column (5) of Table 4 shows that deposit spreads of bank branches located in counties with higher concentrations increase more compared to branches located in areas with lower concentrations, even comparing the branches of the same bank located in the same state. Specifically, the deposit rate increase is around 3 bps (10*0.30=3) less for the bank branch that operates in a county with a 0.30 higher HHI than the county where the other branch is located.

Table 4 Panel B reports the results for \$25K Money Market accounts, a common saving deposit type. It shows that spreads increase around 85 bps after a 100 bps increase in the one-year Treasury yield, which is 25 bps higher for branches located in concentrated counties.³² In other words, the average deposit rate on \$25K Money Market accounts increases only 15 bps after a 100 bps increase in the one-year Treasury yield. Moreover, it increases around 7.5 bps (25*0.30=7.5) less for bank branches that operate in counties with a 0.30 higher HHI. Column (5) of Table 4 shows that deposit spreads of bank branches located in counties with higher concentrations increase more compared to branches located in areas with lower concentrations, even comparing the branches of the same bank located in the same state. As seen from the results, the interest rate pass-through is much lower for savings deposits than time deposits.³³

Tables 5 to 6 present the results for various loan products. The results indicate that loan spreads, $(i^L - i^{UST})$ increase after monetary policy tightening.³⁴ However, monetary policy pass-through to loan spreads is lower in more concentrated areas. One explanation for this result is that branches located in more concentrated areas reduce their loan markups to mitigate the effects of the fall in loan demand without losing profits, as they can increase their deposit spreads from the funding side. Specifically,

³²The magnitude of the rise in the deposit spread is slightly higher than the one reported in Drechsler, Savov, and Schnabl (2017) due to the instrumental variable approach used in my paper. Usage of the Bauer and Swanson (2022) shock slightly amplifies the response of the deposit spread.

³³Time deposits are locked in for a term, whereas checking and saving deposits can be withdrawn immediately.

³⁴Results are robust using the loan rates instead of spreads.

Table 5 Panel A Column (1) show that personal loan spreads increase around 18 bps after a 100 bps increase in the one-year Treasury yield.³⁵ Yet, this increase is around 29 bps less for banks in areas with higher market concentration. That is, loan rates increase around 8.7 bps (29*0.30=8.7) less in bank branches that serve in counties with a 0.30 higher Herfindahl-Hirschman Index. Columns (2) to (5) report the results for the banks that operate at least in two counties for different fixed effect specifications, and the results are robust. Table 5 Panel B reports the results on 72-month automobile loans. The spread on automobile loans increases by 25 bps on average, but it rises 19 bps less for banks in more concentrated areas. If a bank branch is located in a county with a 0.30 higher Herfindahl-Hirschman Index, the rate on automobile loans increases by 5.7 bps (19*0.30=5.7) less on average.³⁶ Column (5) shows the result remains significant even when we compare the branches of the same bank operating in the same state yet located in counties with different degrees of concentration.

Table 6 reports similar results for HELOCs and 30-year fixed mortgage spreads.³⁷ Specifically, Panel A documents that spread on Home Equity Line of Credits (HE-LOCs) with less than 80% loan-to-value ratio (LTV) increases by 97 bps on average, yet it increases approximately 25 bps less for branches located in more concentrated markets.³⁸ Columns (2) to (5) present the results for the banks that operate at least in two counties for various fixed effect specifications, and the results remain robust. Mainly, Column (5) shows that among the branches of the same bank that operates in the same county, the one located in a more concentrated county increases its loan spread less for both products. Panel B reports similar results on 30-year fixed mortgage spreads. In particular, Column (5) shows that the loan spread increases around 33 bps less for branches located in more concentrated markets. That is, if a bank branch is located in a county with a 0.30 higher Herfindahl-Hirschman Index, the loan rate on mortgages increases by 9.9 bps less (33*0.30=9.9) on average.

Taken together, the results indicate that monetary policy pass-through to deposit and loan spreads are lower for bank branches located in more concentrated areas. In particular, these branches widen their deposit spreads more and offer lower deposit rates to their customers. At the same time, they also increase their markups on loans

³⁵I use the personal loan rates for Tier 1 customers, which has the best credit score among all other customers.

³⁶I calculate the automobile loan spread by subtracting the average of 5 and 7-year Treasury yields from the automobile loan rates.

³⁷I report the results on the most responsive loan products for each category.

³⁸HELOCs are loans that allow you to borrow against your home's equity.

less to alleviate the effect of contractionary monetary policy on loan demand.

The previous results document that banks operating in concentrated areas increase their deposit spreads more, allowing them to keep their loan demand more stable by reducing the pass-through of higher policy rates to loan rates. Therefore, if banks that operate in concentrated markets compensate for the decrease in deposit inflows by relying on wholesale funding, as the aggregate data in Section 3 suggest, I expect these banks' lending to decrease relatively less following a contractionary monetary policy. Thus, banks' reliance on wholesale funding can mitigate the impact of deposit outflows and result in less loan contraction for banks with higher market power that operate in highly concentrated local banking markets.

Since banks make their decisions on wholesale funding at the bank level and channel their funds across branches when needed, I test this hypothesis using aggregate balance sheet data from Call Reports in the next section. That is, I explore the impact of bank market power on the transmission of monetary policy on banks' wholesale funding spreads and funding composition. I also provide a complete picture of the underlying mechanism that leads to a lower monetary policy pass-through to loan spreads and lending for banks with higher market power.

5.2 Bank-level Estimation

In this section, I examine the effect of the bank market power on the aggregate banklevel variables to provide a comprehensive picture of the mechanism that diminishes the impact of monetary policy on bank lending. My theory suggests that banks that operate in highly concentrated markets adjust their deposit rates less and compensate for the decrease in their deposit by relying on wholesale funding as they access wholesale funding at a comparatively lower cost. Moreover, they increase the interest rates on their loans relatively less as funding costs increase less for them. Consequently, they dampen the effect of contractionary monetary policy on their lending.

In order to test this mechanism, I construct a bank-level measure of market power, Bank-HHI, by averaging the local concentration of the counties where the bank's branches operate (H_{ct}), weighing each branch by its lagged share of assets, and use this measure as a proxy for the bank market power. This measure specifically allows me to capture the impact of market power on banks' cost of accessing wholesale funding and wholesale funding reliance as decisions related to wholesale financing are conducted at the bank level. In addition, it further allows me to capture the impact of bank market power on lending.

$$BankHHI_{jt} = \sum_{i \in j} \left(\frac{assets_{it-1}}{\sum_{i \in j} assets_{it-1}} \times HHI_{ct} \right)$$
(3)

Equation (3) presents the calculation for the bank market power measure, where HHI_{ct} is the concentration of a particular county where branch *i* of bank *j* is located and $assets_{it-1}$ is the total assets of branch *i* of bank *j*.

To capture the impact of bank market power for the pass-through of monetary policy to bank-level outcomes, I run the following regression at the bank-quarter level:

$$\Delta y_{jt} = \alpha_j + \beta_1 \Delta R_t + \beta_2 \Delta R_t \times BankHHI_{jt-1} + \gamma BankHHI_{jt-1} + \Gamma'(L)X_{jt-1} + \Omega'(L)Z_t + \epsilon_{jt}$$
(4)

Where Δy_{jt} is the log change in a given balance sheet component of bank *j* from date *t*-1 to *t*. ΔR_t is the change in the one-year Treasury yield from *t*-1 to *t* instrumented with Bauer and Swanson (2022) shocks. *BankHHI*_{*jt*-1} is the bank-level concentration of bank *j*, lagged by one period. α_j is bank fixed effects, and X_{jt-1} is bank-level controls such as the lagged change in the banks' assets, equity, and liquidity. Specifically, these bank-level variables enable me to control for differences that may stem from the bank size, liquidity, and bank soundness. They are added in log difference form to capture the time-series trend.³⁹ I also add bank fixed effects interacted with a dummy for the ZLB period to ensure that my results are not driven by the zero lower bound period. In addition, I add the following control variables: GDP growth, unemployment rate, inflation.⁴⁰ These variables are captured by the term Z_t . I cluster the standard errors at the bank level to control for correlation within banks.⁴¹ I demeaned the Bank-HHI and all other control variables.⁴²

Table 7 reports the results. Column (1) of Table 7 Panel A shows that after a 100 bps points increase in the one-year Treasury yield, deposit spreads of banks (measured as the fed funds rate minus domestic deposit interest expense divided by domestic

³⁹Results are robust using lagged values of total assets, equity and liquidity to asset ratios.

⁴⁰Results are robust, adding the change in the commercial paper spread and mortgage spread.

⁴¹My results are robust, including time f.e and reported in Section 7. As suggested in Cameron and Miller (2015), adding time fixed and clustering at the other dimension eliminates concerns on error correlation in multi-dimensions, if any.

⁴²I winsorize all variables at the 1% and 99% level by quarter to isolate the effect of outliers following Drechsler, Savov, and Schnabl (2017).

deposits) increase by 88 bps on average. That is, the average deposit rate banks pay to their depositor's increases around 12 bps after a 100 bps increase in the policy rate. Notably, the increase in the average deposit rates is lower than the amount reported on time and saving deposits presented in the previous section, as it also includes the transaction deposits.⁴³ The interaction term on the change in the policy rate and bank-level concentration indicates that deposit spreads of banks with higher market power increase by 8 bps more, consistent with my branch-level results. That is, banks with higher market power partially pass the increase in interest rate to their depositors. As a result, they experience a slightly higher decrease in their growth of deposits, as reported in Table 7 Panel B Column (1). However, the change in the deposit growth does not significantly differ among banks at the bank level, indicating that these banks face a relatively inelastic supply of deposits due to their market power. ⁴⁴

Table 7 Panel A Column (2) shows that a 100 bps increase in the one-year Treasury yield leads to a 71 bps increase in the wholesale funding spreads (measured as the fed funds rate minus wholesale funding interest expense divided by total wholesale funding). However, banks with higher market power access wholesale funding with a lower cost as the wholesale funding spreads increase around 20 bps more for those banks. Table 7 Panel B Column (4) shows that banks partially compensate for the decrease in total deposit inflows by relying on wholesale funding. It also presents that wholesale funding reliance increases significantly more for banks with higher market power. Table 7 Column (5) indicates that the increase in wholesale funding reliance enables banks with higher market power to entirely offset their shortfalls in deposits, as liabilities do not significantly differ between banks with high versus lower market power. Specifically, the coefficient on the interaction term between the change in the interest rate and market power is positive. Taken together, the results indicate that monetary policy changes the funding composition of the banking sector by creating a dispersion among the funding spreads of the banks with different degrees of market power.

Table 7 Panel A Column (3) displays the results for bank funding spreads, which are calculated as the weighted average spread on deposits and wholesale funding. In particular, funding spreads increase more for banks with higher market power. This,

⁴³Transaction deposits include interest and non-interest-bearing checking deposits, NOW accounts, ATS accounts, and telephone and preauthorized transfer accounts.

⁴⁴My results suggest that the responsiveness of the deposits to change in the deposit spread decreased in my sample period compared to Drechsler, Savov, and Schnabl (2017) who center on the pre-ZLB period.

in turn, allows these banks to increase their loan spreads less to keep their loan demand stable, as shown in Column (4). Specifically, the average loan spread increases by 28 bps after a 100 bps increase in the one-year Treasury yield. However, it increases around 8 bps less for a bank with higher market power. Overall, the results confirm that bank market power mitigates monetary policy transmission to bank funding and lending rates.

Focusing on the impact of bank market power on bank profitability, I find that, on average, the banks' net interest margins decrease following a monetary contraction, as reported in Table 7 Panel A Column (5). Although there is a decline in the bank's net interest margins, the magnitude is small. Specifically, a 100 bps increase in the Federal funds rate decreases net interest margins by around 3 bps for an average bank. This result is consistent with Drechsler, Savov, and Schnabl (2021), who show that banks closely match the interest rate sensitivities of their interest income and expense. On the other hand, I document that net interest margins decrease significantly less for banks with higher market power due to the higher increase in their funding spreads, although the magnitude of the difference is small. Column (6) reports the results on ROA. Notably, bank profits are insensitive to fluctuations in interest rates, and the profits of the banks with higher market power slightly increase more for banks with higher market power.

As banks with higher market power increase their lending spreads less on average and increase wholesale funding reliance substantially more, I expect the lending of banks with higher market power decreases less. Table 8 Columns (1) and (2) show that total assets and loans reduce after policy tightening. However, total assets and loans decrease less for banks with higher market power. Similarly, securities also fall significantly less for banks with higher market power. In section B, I replicate my results using the LP-IV approach of Jordà (2005) and show that the impact of monetary policy on bank-level lending amplifies in longer horizons.

5.3 Bank-County Estimation

In the previous section, I have shown that funding spreads increase significantly more for banks with higher market power following a contractionary monetary policy. Consequently, these banks adjust their loan spreads less to mitigate the impact of contractionary monetary policy on their lending. Moreover, I document that the lower cost of accessing wholesale funding allows banks with higher market power to replace their deposit outflows with wholesale financing. Hence, lending of banks with market power decreases relatively less.

Next, I use branch-level small lending data from FDIC to investigate the link between bank lending and macroeconomic outcomes.⁴⁵ Usage of small lending data is particularly advantageous to establish causality between bank lending and countylevel lending, and further county-level lending and unemployment as local businesses are highly dependent on small business lending to fund themselves. Moreover, bankcounty level small business lending data allows me to control for differences in local lending opportunities and rule out the possibility that my results are driven by the local loan demand. Specifically, the main threat to identification in this setting is that borrowers in different counties might be distinctly affected by the macroeconomic environment resulting from the change in the policy rate. Consequently, a bank lending in a particular county might be influenced by the change in local lending opportunities independent of the bank's market power.

To tackle this issue, I add county-time fixed effects to my analysis that controls for the time-varying changes in the local loan demand. In addition, as banks execute wholesale funding decisions at the bank level and can allocate funds internally across their branches to fund their lending if needed, I use my bank-level concentration measure in my analysis which is a good indicator of how much funding a bank will raise and increase its profits from the funding side. In particular, if banks with higher market power can compensate for their shortfalls in deposits through cheaper access to wholesale funding and partially pass the increase in the policy rate to their borrowers, these banks' lending should decrease less compared to other banks.

To test this hypothesis, I estimate the following regression at the bank-county level:

$$\Delta y_{jct} = \alpha_{jc} + \beta_1 \Delta R_t + \beta_2 \Delta R_t \times BankHHI_{jt-1} + \gamma BankHHI_{jt-1} + \Omega'(L)Z_t + \epsilon_{jt}$$
(5)

 Δy_{jct} is the percentage change in the small business lending by bank *j* in county *c* from year *t*-1 to *t*.⁴⁶ *BankHHI*_{*jt*-1} is the bank-level concentration of bank *j* in year *t*-1. ΔR_t is the change in the one-year Treasury yield from *t*-1 to *t* instrumented with Bauer and Swanson (2022) shocks. α_{jc} are bank-county fixed effects that absorb time-

⁴⁵I exclude 2008 from my analysis as small business lending decreased around 30 percent during this period due to the adverse effect of the financial crises.

⁴⁶I use percentage change in loans rather than the level of loans to be able to account for differences in bank size, which considerably impacts banks' loan volumes.

invariant characteristics such as banks' brand effects.⁴⁷ This approach allows me to capture the average effect of an increase in the interest rate on bank lending and identify the impact of bank market power on the transmission of monetary policy to bank lending. I further add county-time fixed effects, which soaks up the changes in local lending opportunities into my regression. Although including county-time fixed effects is preferable because it isolates the effects that may stem from the change in local lending opportunities, it requires excluding the ΔR_t , which captures the average effect of monetary policy on bank lending outcomes. Hence, I report the results obtained through both approaches and show that my results are robust to different specifications. Z_t includes GDP growth, unemployment rate, inflation.⁴⁸ These controls are added into the regression to isolate the role of the level of interest rates from that of cyclical conditions whenever time or county-time fixed effects are excluded. Standard errors are double clustered at the bank and county levels.

Table 8 displays the results on bank-county level lending. Column (1) presents the results across banks and regions. Specifically, it shows that bank-level small business lending decreases approximately by 14 bps after a 100 bps increase in the one-year Treasury yield, yet lending of banks with higher market power falls around 29 bps less in line with my hypothesis. In particular, small business lending decreases by 8.7 bps (28*0.30=8.7) less for banks with a 0.30 higher Bank-HHI. Column (2) includes the local concentration of the counties to the regression, and the interaction term between the change in the policy indicator and Bank-HHI remains economically and statistically significant. This result confirms that the bank-level market power rather than the county-level market power drives the wholesale funding channel. Column (3) adds time-fixed effects, and the interaction term coefficient remains significant. Column (4) repeats the same analysis by controlling for the local concentration of the counties and reports similar results. Finally, Column (5) presents the results with county-time fixed effects that absorb the impact of loan demand. The magnitude of the coefficient on the interaction term remains similar to Columns (1) and (2); additionally, it is still significant.

The results confirm that market power enables banks to increase their funding spreads, and banks with higher market power offset the higher decrease in deposit outflows by increasing their wholesale financing. This also allows banks with higher to increase

⁴⁷County fixed effects interacted with a dummy variable for the zero-lower bound period is also added to control for differences that may stem from the ZLB period.

⁴⁸Results are robust adding the change in the commercial paper spread, and 30-year mortgage spread.

their loan rates less on average. Consequently, their lending decreases relatively less after Fed raises the policy rate. In the next section, I aggregate my data to the county level to show that my results on bank-level lending have implications for the transmission of monetary policy to county-level small business lending and real economic outcomes.

5.4 County-level Analysis

In the previous section, I've shown that small business lending of the banks with higher market power decreases less compared to other banks. This result suggests that total lending in counties where banks with higher market power operate should decrease less after an increase in the policy rate. Thus, real economic outcomes such as unemployment should be affected less negatively in these regions as small business lending constitutes a substantial amount of the funding of the local businesses.

To test this prediction, I aggregate my bank-level small business lending data at the county level and construct a county-level concentration measure, County-HHI, defined as the weighted average of Bank-HHI across all banks lending in a given county, using their lagged lending shares as weights to alleviate concerns regarding the endogeneity of the measure.

$$CountyHHI_{ct} = \sum_{j \in c} \left(\frac{lending_{jct-1}}{\sum_{j \in c} lending_{jct-1}} \times BankHHI_{jt} \right)$$
(6)

In particular, County-HHI measures the extent to which a county is served by banks with higher market power and allows me to test the impact of bank market power on county-level outcomes.

As county-level macroeconomic outcomes such as unemployment might respond to monetary policy with a lag, I use the local projection (LP-IV) method of Jordà (2005) which allows me to capture the impact of bank market power in longer horizons. Equation (7) presents the baseline LP-IV specification, which estimates the dynamic causal effects of monetary policy changes subject to the alternative banking concentration of the counties.

$$\Delta_{h} log(y_{ct+h}) = \alpha_{c}^{h} + \beta^{h} \Delta R_{t} + \Gamma^{h} \Delta R_{t} \times CountyHHI_{ct-1} + \theta^{h}CountyHHI_{ct-1} + \Omega'(L)Z_{t} + \epsilon_{ct+h}$$
(7)

Where the horizon is h = 0, 1, ...4 years, and c and t denote county and time, respectively. The left-hand side of equation (7) is the cumulative change in the outcome variable y, $\Delta_h log(y_{c,t+h}) = log(y_{c,t+h}) - log(y_{c,t-1})$, where y is the total county-level small business lending or unemployment.⁴⁹ The specification regresses the dynamic cumulative change in variable y on monetary policy changes subject to the banking concentration of counties. α_c^h denotes county fixed effect, which absorbs permanent differences across counties.⁵⁰ ΔR_t refers to the change in the one-year Treasury yield from year t-1 to t instrumented with Bauer and Swanson (2022) shocks. β^h captures the impulse response of the left-hand side variable at time t+h to a monetary policy change at time t. Γ^h gives the marginal effect of concentration of Bauer and Swanson (2022) shocks and the County-HHI variable.⁵¹

 Z_t includes the following control variables: change in the national level GDP, unemployment rate, inflation.⁵² Standard errors are clustered by county level to control for correlation within counties. The estimation is calculated up to a horizon of four years, and the lag structure on all right-hand-side variables is one year.

Figure 7 plots the estimated coefficients as well as their 95% confidence intervals on the total county-level small business lending. Figure 7a shows that total lending decreases by around 10 bps percent after a 100 bps increase in the policy rate. Moreover, the peak effect reaches 40 bps four years after the contractionary monetary policy. In contrast, Figure 7b shows sizable heterogeneity in monetary policy outcomes conditional on the County-HHI of the counties. Specifically, county-level lending decreases around 10 bps less in counties in which banks with higher credit market power operate. The peak effect is about 50 bps, occurring three years after the monetary policy

⁴⁹Using the level of loans as a left-hand side variable is particularly problematic in this case as the marginal effects implied by the level specification are implausible for any cross-section of markets that vary substantially in size. A change in the federal funds rate will have a much greater effect on loan volume in a large market than in a small market, as argued by Adams and Amel (2011).

⁵⁰I further include county fixed effects interacted with a dummy variable for the ZLB period to control the effect of the low interest rate environment.

⁵¹I further add the local concentration of the county interacted with the change in the policy instrument.

⁵²Results are robust adding mortgage and commercial paper spread.

shock. Figure 7c plots the results for counties with County-HHI 0, 0.15, 0.20 and 0.30, respectively.⁵³ As shown in the Figure, county-level small business lending decreases around 10 bps following a 100 bps increase in the one-year Treasury yield. However, it reduces by around 3 bps less in counties with a County-HHI of 0.30. These findings highlight that small business lending in counties where banks with higher market power operate is less negatively affected by contractionary monetary policy relative to other counties. As small businesses are highly dependent on local lending and have a sizeable effect on the county's real economy, unemployment should also increase less in counties served by banks with higher market power following an increase in the policy rate.

Figure 8 presents the results on unemployment. Specifically, Figure 8a shows that the increase in unemployment reaches up to 15 bps after a 100 bps increase in the one-year Treasury yield. Figure 8b suggests that the unemployment rate rises less in counties with higher County-HHI, and it is more sizeable starting from a year after the interest rate hike. Overall, the results provide evidence that market power alleviates the negative impact of interest rate hikes on county-level lending and unemployment. Hence, market power has a crucial effect on the transmission of monetary policy to real economic activity.

The following section presents a partial equilibrium model of monopolistic competition that rationalizes my findings on the partial pass-through of monetary policy to the bank deposit and lending rates for the banks with higher market power.

6 A Simple Model of Monopolistic Competition

To provide intuition for the underlying mechanism and rationalize my empirical findings, I build a simple model of monopolistic competition and take it to the data. The model allows me to look at the simultaneous exercise of market power on both the deposit and lending side-necessary to understand previous empirical results on the lending channel for monetary policy.

The model assumes that deposits and loans are baskets of differentiated products with constant elasticity of substitution, which leads to a constant markup on the retail rates. The building block of the model comes from Ulate (2021) where the deposit supply and loan demand for each bank rise from the fact that depositors and borrow-

⁵³These values correspond to the 25th, 50th, and 75th percentile of the County-HHI distribution.

ers have CES preferences across banks. The model takes the aggregate amounts of deposits supplied, and loans demanded as given since this is a partial equilibrium exercise. It predicts that pass-through to loan and deposit rates decrease for banks with higher market power which access the wholesale funding markets at a cheaper cost. That is, the model suggests that pass-through to the retail rates is determined by the cost of accessing wholesale funding, which is a function of bank market power.

6.1 The Model:

Assume that banks' cost of accessing wholesale funding is exogenously determined, and banks with higher market power access wholesale funding markets at a lower cost, consistent with the data. Banks operate under monopolistic competition, where market power could arise from product differentiation. Table 1 shows a bank's balance sheet with loans, L_j , securities G_j , as assets; and deposits D_j and wholesale funding WF_j , as liabilities.

Table 1: Balance Sheet

Assets	Liabilities
Loans <i>L_j</i>	Deposits D_j
Securities G_j	Wholesale Funding WF_j

Each bank *j* maximize profits given by equation (8):

$$\max_{i_j^L, i_j^D} \Pi_j = (1+i_j^L)L_j + (1+f)G_j - (1+i_j^D)D_j - \left(1+f - \phi_j^{WF}\right)WF_j$$
(8)

subject to the loan and deposit demand and the bank balance sheet constraint given by

$$L_j = \left(\frac{1+i_j^L}{1+i^L}\right)^{-\theta^\ell} L \tag{9}$$

$$D_j = \left(\frac{1+i_j^D}{1+i^D}\right)^{-\theta^a} D \tag{10}$$

$$L_j + G_j = D_j + WF_j \tag{11}$$

Where ϕ_j^{WF} is the wholesale funding spread of the banks, assumed to be exogenously higher for banks with higher market power as wholesale funding is cheaper

for them.⁵⁴ $1 < -\theta^{\ell}$ is the elasticity of substitution for loans between banks, and L is the aggregate loan in the economy. i^L is the aggregate loan rate index. $\theta^d < -1$ is the elasticity of substitution for deposits between banks, D is the aggregate deposit in the economy, and i^{D} is the aggregate deposit rate index. The loan and deposit demand functions are derived by solving the saver and the borrower problems, where both agents have CES demand functions. These demand functions are driven in Appendix C.

The maximization problem of the bank can be solved by substituting equations (9) to (11) into equation (8) and taking first-order conditions with respect to deposit and loan rates. The solution of the problem yields the loan and deposit rates as markup and markdown over the federal funds rate and the bank's wholesale funding spread, ϕ_i^{WF} .55

$$1 + i_j^L = \underbrace{\frac{\theta^\ell}{\theta^\ell - 1}}_{\text{markup}} \left(1 + f - \phi_j^{WF} \right) \tag{12}$$

$$1 + i_j^D = \underbrace{\frac{\theta^d}{\theta^d - 1}}_{\text{markdown}} \left(1 + f - \phi_j^{WF} \right)$$
(13)

As $1 < \theta^{\ell}$ and $\theta^{d} < -1$ which indicates that

$$egin{aligned} & rac{ heta^\ell}{ heta^\ell-1} > 1 \ & 0 < rac{ heta^d}{ heta^d-1} < 1 \end{aligned}$$

Equation (12) shows that the loan rate is a markup over the federal funds rate and the bank's wholesale-funding spread, ϕ_i^{WF} . On the other hand, equation (13) shows that the deposit rate is a markdown over the federal funds rate and the bank's wholesalefunding spread, ϕ_i^{WF} . Since wholesale funding spreads, ϕ_i^{WF} are higher for banks with higher market power, both equations (12) and (13) indicate that deposit and loan rates are lower for banks with higher market power. In order to capture the impact of market power on banks' wholesale funding, deposit and loan volumes, we can now turn to equations (9), (10), and (11). In particular, equation (9) suggests that the loan demand of the banks with higher market power is higher as these banks offer lower

⁵⁴Note that ϕ_j^{WF} is assumed to be a function of the federal funds rate and widens more for banks with higher market power after policy tightening in line with the data. ⁵⁵Note that higher ϕ_j^{WF} indicates that the bank accesses wholesale funding with a lower cost.

loan rates. On the other hand, equation (10) indicates that the deposit supply of banks with higher market power is lower due to the lower deposit rate they offer. Hence, banks with higher market power rely more on wholesale funding, which can be seen through equation (11).⁵⁶ Overall, the results show that the model matches with the data in terms of the response of deposit and loan rates as well as the response of banks' funding and lending volumes.

The paper's main focus is the monetary policy pass-through to deposit and loan rates; thus, I differentiate the loan and deposit rate equations with respect to the policy rate, f. Equations (14) and (15) suggest that monetary policy transmission to deposit and loan rates depend on pass-through to banks' wholesale funding spreads. As $\frac{d\phi_j^{WF}}{df}$ is higher for banks with higher market power, in other words, wholesale funding spreads in the policy rate, pass-through to retail rates is lower for these banks. That is, banks with higher market power increase both the loan and the deposit rates less after an increase in the policy rate, verifying the predictions of my empirical analysis.

$$\frac{di_j^L}{df} = \frac{\theta^\ell}{\theta^\ell - 1} - \frac{\theta^\ell}{\theta^\ell - 1} \frac{d\phi_j^{WF}}{df}$$
(14)

$$\frac{di_j^D}{df} = \frac{\theta^d}{\theta^d - 1} - \frac{\theta^d}{\theta^d - 1} \frac{d\phi_j^{WF}}{df}$$
(15)

The next section test model predictions by taking the model into data. That is it shows that:

- 1. Monetary policy pass-through to loan rates decreases with market power.
- 2. Monetary policy pass-through to deposit rates decreases with market power.

6.2 Model Assessment

In this section, I test whether the model performs well in matching the data by using data from the Call Reports .

First, I estimate the average markup and markdown of the U.S banking system by using the following equations, which simply assume that the loan and deposit rate is

⁵⁶Banks use any extra funding to issue securities as also suggested by equation (11).

a constant markup/markdown over the federal funds rate.

$$1 + i^{L} = \underbrace{\frac{\theta^{\ell}}{\theta^{\ell} - 1}}_{\text{markup}} (1 + f)$$
(16)

$$1 + i^{D} = \underbrace{\frac{\theta^{d}}{\theta^{d} - 1}}_{\text{markdown}} (1 + f)$$
(17)

Where i^L is the average loan rate for all banks, i^D is the average deposit rate for all banks from the Call Reports, and f is the federal funds rate. The average markup for loans is found as 2.47, whereas the average markup on deposits is found to be 1.17, which is slightly higher than 1 as the deposit rates exceed the federal funds rate from time to time, as shown in Figures 3a and 3c.⁵⁷

Next, I cross-validate the model to find the model implied loan and deposit rates. To do so, I obtain the average change in the wholesale funding spread, $\Delta \phi^{WF}(WF)$ from the Call Reports using equation (18). I then plug it into equations (19) and (20). I perform this exercise both for the banks above and below the 75th percentile of Bank-HHI distribution.

$$\Delta(1+i^{WF}) = \Delta(1+f-\phi^{WF}) \to \Delta\phi^{WF}(WF)$$
(18)

$$\Delta \left[1 + i^L \right] = \Delta \left[\frac{\theta^\ell}{\theta^\ell - 1} \left(1 + f - \phi^{WF}(WF) \right) \right] \to \Delta i^L$$
(19)

$$\Delta \left[1 + i^D \right] = \Delta \left[\frac{\theta^d}{\theta^d - 1} \left(1 + f - \phi^{WF}(WF) \right) \right] \to \Delta i^D$$
(20)

Figure 9a presents the results for the change in the loan rate, where the left-handside panel reports the model implied change in the loan rate and the right-hand side is the actual change in the loan rate from the Call Reports, where the average loan rate for the High-HHI group is calculated by averaging the loan rate of all banks that are above the 75th percentile of the Bank-HHI distribution for a given year. Again, both the replications using the model and the actual data suggest that the loan rate changes less for the banks with higher market power. Similarly, Figure 9b reports the results

⁵⁷Note that I averaged these markups/markdowns over time.

for the deposit rate. It shows that both the actual data and the model indicate that the deposit rate changes less for the banks with higher market power. These exercises show that the simple model performs well in matching the data as monetary policy pass-through to deposit and loan rates decreasing with market power.

7 Robustness

This section conducts a large number of checks that confirm the baseline results are robust to alternative estimation strategies, usage of different market power measures, monetary policy shocks, and samples.

Usage of Alternative Loan and Deposit Products

I confirm the robustness of my results using the alternative loan and deposit products. In particular, I add 6-month certificates of deposits, 10K money market funds and 15year fixed-rate mortgages to my analysis. Table B.1 and B.2 in Appendix report the results and indicates that banks that operate in more concentrated areas increase their deposit spreads more, whereas their lending spreads less, verifying my main findings. **Usage of an Alternative Concentration Measure**

Figure 6a and Figure 6b show that the loan and deposit market powers are highly correlated. Hence, both measures provide a good proxy for bank concentration. I re-estimate Equation (2) using the deposit market power measure to confirm that my results are robust using both concentration measures. Table B.3 shows that deposit spreads widen more whereas loan spreads less in more concentrated banking regions, consistent with my main results.

Estimation of Bank-level Results using LP-IV strategy

In general, funding and lending rates respond to monetary policy more rapidly. On the other hand, the transmission of monetary policy to bank-level assets and liabilities may take time. To address this issue, I repeat my analysis on bank-level balance sheet variables using the LP-IV approach of Jordà (2005). The LP-IV strategy enables me to capture the impact of monetary policy on bank-level variables over more extended periods. Equation (21) presents my estimation strategy:

$$\Delta_{h}log(y_{jt+h}) = \alpha_{j}^{h} + \beta^{h}\Delta R_{t} + \Gamma^{h}\Delta R_{t} \times BankHHI_{jt-1} + \theta^{h}BankHHI_{jt-1} + \Gamma'(L)X_{jt-1} + \Omega'(L)Z_{t} + \epsilon_{jt+h}$$
(21)

Where the horizon is h = 0, 1, ...8 quarters, and j and t denote bank and time, respectively. The left-hand side of equation (21) is the cumulative change in the bank-balance sheet variable y calculated as: $\Delta_h log(y_{ct+h}) = log(y_{ct+h}) - log(y_{ct-1})$. ΔR_t refers to the instrumented change in the one-year Treasury yield. BankHHI_{it-1} is the lagged banklevel concentration of bank *j* given by equation (3). X_{jt-1} includes the lagged change in the bank-level assets, liquidity, and equity. Z_t includes the following control variables: change in the national level GDP, unemployment rate, inflation.⁵⁸ The lag structure on the control variable is set to be two quarters, and standard errors are clustered at the bank level.⁵⁹ Figures B.1 to B.10 in Appendix plot the response of bank-level variables to monetary policy conditional on bank market power. Figures B.1 to B.3 show that my results hold using funding and lending rates rather than spreads. The Figures reveal that interest pass-through to bank retail and wholesale funding rates decreases even more for the banks with higher market power over time. Figure B.9 present that wholesale funding reliance is much higher for the banks with higher market power after two quarters of the monetary policy contraction. Figures B.5 and B.6 suggest that both bank-level assets and loans decrease less for banks with higher market power, where the effect becomes notable two years after the monetary policy shock, at the time the impact on wholesale-funding starts to amplify, verifying my baseline findings. For other bank-balance sheet variables, I also obtain results similar to the ones reported in Table 7 and confirm that my results are robust to alternative specifications. Usage of Additional of Bank-level Controls

I also test whether heterogeneity in other observable bank characteristics, such as bank size, can drive my main results. To do so, I re-estimate the main results using the specification in Equation (4), where monetary shocks are interacted with various bank characteristics. Specifically, I interact with the monetary policy with bank size, equity, and liquidity measures to ensure that my results are not driven by the differences in bank characteristics, especially the differences in bank size. Table B.6 in Appendix report the results. In each case, the coefficient on monetary policy and bank concentration interaction remains similar to the reported in the base line specification, suggesting that the main results are not driven by bank size, leverage, or liquidity differences across the banks.

Usage of the Time Fixed Effect Specification

I test the robustness of my results by adding time fixed effects to my analysis on bank

⁵⁸Results are robust including the change in the commercial paper spread, and change in the mortgage spread.

 $^{^{59}}$ I also include a dummy for the ZLB period to control for the changes this may stem from this period.

balance sheet variables as in Drechsler, Savov, and Schnabl (2017). Tables B.7 in Appendix present that results are robust using either approaches.

Usage of Alternative Monetary Policy Shocks

I checked the robustness of my results by using the pure change in the one-year Treasury yield and orthogonalized monetary policy shocks from Bauer and Swanson (2022) and obtained similar results. In particular, usage of orthogonalized Bauer and Swanson (2022) shocks addresses the concerns on the potential correlation between monetary policy surprises and macroeconomic or financial data that becomes publicly available before the FOMC announcements. These shocks are obtained through regressing the standard Bauer and Swanson (2022) shocks, constructed by the first principal component analysis to ED1-ED4, on the economic and financial variables that predate the announcements and then taking the residuals. Tables B.4 and B.5 in Appendix re-estimate results on deposit and loan spreads with both the pure change in the one-year Treasury yield and orthogonalized Bauer and Swanson (2022) shocks. The findings confirm the baseline findings, although the magnitude of the coefficients slightly increases for the specification with orthogonalized shocks.

Usage of Full Small Lending Sample:

I verify the robustness of my result by using the full sample on bank-county level small business lending. That is, I include the loan originations less than \$100,000 in value in my sample. Table B.8 in Appendix reports the results. As seen from the table, lending of banks with higher market power decreases less compared to other banks, consistent with my main findings in Table 8.

8 Conclusion

In this paper, I study the importance of bank market power for the monetary policy pass-through to bank funding dynamics, lending, and profitability. First, I document that monetary policy creates a considerable variation in banks' funding spreads depending on banks' market power. I find that both the wholesale funding and deposit spreads increase more for banks with higher market power after a policy tightening, and the funding composition of these banks changes substantially. In particular, I show that deposit inflows of banks with higher market power decrease slightly more due to a lower pass-through of the increase in the policy rate to their deposit rates, and these banks compensate for their funding shortfalls through wholesale funding as they access wholesale financing at a lower cost. Moreover, I document that the rise in

funding spreads enables banks with higher market power to alter their loan rates less and smooth their lending and profitability. I further report that bank market power affects monetary policy transmission to the macroeconomy by showing that countylevel lending and employment are less adversely affected by the monetary contraction in regions where banks with higher market power operate. Lastly, I build a theoretical model featuring monopolistic competition and rationalize my empirical findings.

The findings of this paper are crucial for the following reasons. Firstly, this study is the first paper evaluating the impact of bank market power on the transmission of monetary policy to banks' interest spreads, lending, and profitability, considering the interdependence among the deposit, wholesale funding, and credit markets which is crucial to achieving clear understanding on monetary policy transmission mechanism. Importantly, this paper provides a complete picture of the role of bank market power on the monetary policy transmission to bank-level outcomes by revealing the importance of the wholesale funding channel. Finally, the results of this paper have significant implications for policy-making as it presents new insights into the effect of market power on the pass-through of monetary policy to real economic outcomes.

References

- Adams, Robert M and Dean F Amel (2011). "Market structure and the pass-through of the federal funds rate". In: *Journal of Banking & Finance* 35.5, pp. 1087–1096.
- Balloch, Cynthia and Yann Koby (2019). "Low rates and bank loan supply: Theory and evidence from Japan". In: *Unpublished manuscript Brown University* 9.10.
- Bauer, Michael D and Eric T Swanson (2022). *A reassessment of monetary policy surprises and high-frequency identification*. Tech. rep. National Bureau of Economic Research.
- Bernanke, Ben S. (1983). "Nonmonetary Effects of the Financial Crisis in the Propagation of the Great Depression". In: *The American Economic Review* 73.3, pp. 257–276.
- Bernanke, Ben S and Alan S Blinder (1988). "Credit, money, and aggregate demand". In: *The American Economic Review* 78.2, pp. 435–439.
- Bord, Vitaly, Victoria Ivashina, and Ryan Taliaferro (2015). "Large banks and the transmission of financial shocks". In: *Available at SSRN*.
- Brunnermeier, Markus K and Yann Koby (2018). *The reversal interest rate*. Tech. rep. National Bureau of Economic Research.
- Cameron, A Colin and Douglas L Miller (2015). "A practitioner's guide to clusterrobust inference". In: *Journal of human resources* 50.2, pp. 317–372.
- Choi, Dong Beom and Hyun-Soo Choi (2021). "The effect of monetary policy on bank wholesale funding". In: *Management Science* 67.1, pp. 388–416.
- Drechsler, Itamar, Alexi Savov, and Philipp Schnabl (2017). "The deposits channel of monetary policy". In: *The Quarterly Journal of Economics* 132.4, pp. 1819–1876.
- (2021). "Banking on deposits: Maturity transformation without interest rate risk".
 In: *The Journal of Finance* 76.3, pp. 1091–1143.
- Eggertsson, Gauti B, Ragnar E Juelsrud, Lawrence H Summers, and Ella Getz Wold (2019). *Negative nominal interest rates and the bank lending channel*. Tech. rep. National Bureau of Economic Research.
- Gertler, Mark and Peter Karadi (2015). "Monetary policy surprises, credit costs, and economic activity". In: *American Economic Journal: Macroeconomics* 7.1, pp. 44–76.
- Gürkaynak, Refet S, Brian P Sack, and Eric T Swanson (2005). "Do actions speak louder than words? The response of asset prices to monetary policy actions and statements". In: *International Journal of Central Banking* 1.1, pp. 55–93.
- (2007). "Market-based measures of monetary policy expectations". In: *Journal of Business & Economic Statistics* 25.2, pp. 201–212.

- Heider, Florian, Farzad Saidi, and Glenn Schepens (2019). "Life below zero: Bank lending under negative policy rates". In: *The Review of Financial Studies* 32.10, pp. 3728– 3761.
- Jordà, Òscar (2005). "Estimation and inference of impulse responses by local projections". In: *American economic review* 95.1, pp. 161–182.
- Kashyap, Anil K and Jeremy C Stein (1994). "Monetary policy and bank lending". In: *Monetary policy*. The University of Chicago Press, pp. 221–261.
- (2000). "What do a million observations on banks say about the transmission of monetary policy?" In: *American Economic Review* 90.3, pp. 407–428.
- Miranda-Agrippino, Silvia and Giovanni Ricco (2021). "The transmission of monetary policy shocks". In: *American Economic Journal: Macroeconomics* 13.3, pp. 74–107.
- Romer, Christina D, David H Romer, Stephen M Goldfeld, and Benjamin M Friedman (1990). "New evidence on the monetary transmission mechanism". In: *Brookings papers on economic activity* 1990.1, pp. 149–213.
- Scharfstein, David and Adi Sunderam (2016). "Market power in mortgage lending and the transmission of monetary policy". In: *Unpublished working paper. Harvard University* 2.
- Ulate, Mauricio (2021). "Going Negative at the Zero Lower Bound: The Effects of Negative Nominal Interest Rates". In: *American Economic Review* 111.1, pp. 1–40.
- Wang, Yifei, Toni M Whited, Yufeng Wu, and Kairong Xiao (2022). "Bank market power and monetary policy transmission: Evidence from a structural estimation". In: *The Journal of Finance* 77.4, pp. 2093–2141.
- Xiao, Kairong (2020). "Monetary transmission through shadow banks". In: *The Review* of Financial Studies 33.6, pp. 2379–2420.

Figure 1: The U.S Financial Markets

Figures (a) and (b) plot the recent economic phenomenons of U.S financial markets. Figure (a) plots the asset share of the top 5 banks, whereas Figure (b) plots the total assets of Money Market Funds over time. The data are from the U.S. Call Reports and FRED, covering 1994 to 2019.



Figure 2: Outline of the Monetary Policy Transmission Mechanism to Bank Lending

Figures (a) and (b) outline the Monetary Policy Transmission Mechanism to Bank Lending. Figure (a) illustrates the average effect of the monetary policy, whereas Figure (b) illustrates the effect of market power on monetary policy transmission to bank lending.

(a) Average Effect of the Monetary Policy

(b) Effect for the Banks with Higher Market Power



Figure 3: Bank Retail Rates, Profitability and Monetary Policy

Figures (a) and (b) plot the commercial banking sector's average deposit, wholesale funding, and loan rate. The data are from the U.S. Call Reports covering 2000 to 2019. Figure (c) plots the deposit rate on the most widely-offered deposit products using RateWatch data from 2000 to 2019. Lastly, Figure (d) plots the profitability of the U.S. banking system over time.



(c) Average Deposit Rate by Product





Figure 4: Retail Deposits, Wholesale Funding, MMFs and Federal Funds Rate

Figure (a) plots the time series of the federal funds rate against the change in the aggregate amount of retail deposits and the aggregate amount of money market funds. Similarly, Figure (b) plots the wholesale funding to retail deposits ratio for the U.S. commercial banks.



Figure 5: Composition of Wholesale Funding and U.S. Small Business Lending

Figure (a) plots the components of the wholesale funding for U.S. commercial banks. The data source is U.S. Call Reports, covering the period between 1997 to 2019. Figure (b) plots the time series of bank lending to small businesses: The red line plots the total volume of loans in billions of dollars, and the blue line plots the number of new loans in billions. Data is from FDIC and covers between 1997 to 2019.



Figure 6: U.S. Banking Market Concentration

This map shows the average Herfindahl index for each U.S. county. The Herfindahl Hirschman Index is calculated each year using the asset market shares of all banks with branches in a given county and then averaged over the period from 1994 to 2019 for local credit market concentration. On the other hand, Figure (b) reports the Herfindahl Hirschman Index calculated using the deposit market shares of all banks with branches in a given county and then averaged over time. The data source is FDIC.

(a) Local Market Concentration Using Asset Shares

(b) Local Market Concentration Using Deposit Shares



Panel A: Deposit Spread	ls					
	All		Low H	HI	High HHI	
	Mean	Std	Mean	Std	Mean	Std
Δ12MCDspread	0.02	0.25	0.02	0.25	0.01	0.25
$\Delta 06 MCDs pread$	0.01	0.26	0.01	0.25	0.00	0.26
Δ MM25Kspread	-0.01	0.36	-0.02	0.35	-0.03	0.37
Δ MM10Kspread	-0.01	0.37	0.00	0.36	-0.01	0.37
HHI _c	0.18	0.12	0.10	0.03	0.25	0.12
Obs.(branch×quarter)	513,43	7	256,24	15	257,19	92
Panel B: Personal Loan S	Spread					
-	All		Low H	HI .	High H	IHI
	Mean	Std	Mean	Std	Mean	Std
∆Personal Loan Sprd	0.04	1.17	0.04	1.16	0.03	1.18
HHI _c	0.24	0.11	0.16	0.03	0.32	0.12
Obs.(branch×quarter)	162,17	′ 3	81,05	0	81,12	.3
Panel C: Auto Loan Spre	ead					
_	All		Low H	HI	High H	IHI
	Mean	Std	Mean	Std	Mean	Std
ΔAuto Loan Sprd	0.01	0.49	0.01	0.50	0.01	0.48
HHI _c	0.24	0.11	0.16	0.03	0.32	0.12
Obs.(branch×quarter)	76,695	5	38,28	9	38,40	6
Panel D: Mortgage Spre	ads					
_	All		Low H	HI	High H	IHI
	Mean	Std	Mean	Std	Mean	Std
Δ15Yr Mtg Sprd	-0.00	0.35	-0.01	0.34	-0.01	0.36
Δ30Yr Mtg Sprd	-0.02	0.31	-0.03	0.30	-0.02	0.31
HHI _c	0.24	0.10	0.16	0.03	0.31	0.10
Obs.(branch×quarter)	39,554	1	19,76	6	19,78	8
Panel E: HELOC Spread	S					
	A	11	Low I	HHI	High H	IHI
	Mean	Std	Mean	Std	Mean	Std
Δ %80 LTV HELOC Sprd	-0.06	0.66	-0.06	0.68	-0.06	0.64
HHI _c	0.23	0.11	0.16	0.03	0.30	0.10
Obs.(branch×quarter)	143,5	330	71,5	97	71,73	33

Table 2: Summary Statistics for Deposit and Loan Spreads-Ratewatch

Panel A: Bank Le	Panel A: Bank Level Interest Spreads							
		All		Low HHI		Hig	h HHI	
	-	Mean	Std	Mear	n Std	Mean	Std	
∆Deposit sprd		0.02	0.40	0.02	2 0.39	0.02	0.41	
∆Loan sprd		-0.00	0.77	-0.0	0 0.77	-0.00	0.77	
∆Wholesale-fundi	ng sprd	0.00	1.46	0.0	1 1.41	0.00	1.50	
Bank-HHI		0.24	0.13	0.1	6 0.03	0.33	0.13	
Observations		455,487	7	2	11,492	21	1 <i>,</i> 501	
Panel B: Small Bu	isiness Lei	nding						
_	Al	1	L	low HH	II	High HHI		
	Mean	Std	Me	an	Std	Mean	Std	
$\Delta Log(Lending)$	0.07	1.16	0.	06	1.18	0.08	1.14	
Bank-HHI	0.27	0.17	0.	18	0.03	0.37	0.21	
Observations	921,2	233		460,612			460,621	
Panel C: County I	Level Char	acteristics						
	I	A11	-	Low HHI		High	HHI	
	Mean	Std	Me	ean	Std	Mean	Std	
Population	90,109	293,227	148,2	798 3	92,3864	31,457	105,871	
Gdp (in mill \$)	4,959	22,257	6,2	722	27,204	3,387	15,656	
Wages (in mill \$)	2,067	9,343	2,0	695	11,082	1,440	7,140	
Unemp. rate	5.75	1.75	5	5.71	1.74	5.78	1.77	
County-HHI	0.240	0.05	0.2	201	0.02	0.280	0.05	
Obs. (counties)	3,	219		1,510		1,70)9	

Table 3: Summary Statistics for Bank and County Level Variables

This table provides summary statistics on bank-level interest spreads, bank-county level small business lending, county-level lending, GDP, unemployment and wages. In addition, it provides a breakdown by high and low Herfindahl (HHI) using the median HHI for the respective sample. The underlying data is from the FDIC for lending, the U.S. Bureau of Economic Analysis for GDP, and the Bureau of Labor Statistics for other variables. It covers January 2000 to December 2019. GDP data is available starting from 2001.

Panel A: 12-Mon	th CD				
		ΔD	eposit Spread		
	All		\geq 2 Cou	nties	
	(1)	(2)	(3)	(4)	(5)
ΔR_t	0.342***	0.320***			
$\Delta R_t \times HHI_c$	(0.00432) 0.135*** (0.0295)	(0.00524) 0.230*** (0.0400)	0.244***	0.0892**	0.103^{***}
D	(0.0293)	(0.0409)	(0.0353)	(0.0382)	(0.0378)
Branch f.e.	Y	Y	Y	Y	Y
County f.e.	Y	Y	Y	Y	Y
Bank-time f.e.	Ν	Ν	Ν	Y	Y
State-time f.e.	Ν	Ν	Y	Ν	Y
Macro controls	Y	Y	Ν	Ν	Ν
Observations	475,942	205,920	205,920	205,920	205,920
R-squared	0.346	0.315	0.002	0.000	0.000
Panel B. 25K Mo	nov Markot F	unde			

Table 4: Time and Saving Deposit Spreads

Panel B: 25K Money Market Funds

	Δ Deposit Spread					
	All		\geq 2 Cou	nties		
	(1)	(2)	(3)	(4)	(5)	
ΔR_t	0.855***	0.831***				
$\Delta R_t \times HHI_c$	(0.00421) 0.246^{***} (0.0306)	(0.00565) 0.215^{***} (0.0447)	0.302*** (0.0443)	0.226*** (0.0506)	0.250*** (0.0498)	
Branch f.e.	Ŷ	Y	Ŷ	Ŷ	Ŷ	
County f.e.	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	
Bank-time f.e.	Ν	Ν	Ν	Y	Y	
State-time f.e.	Ν	Ν	Y	Ν	Y	
Macro controls	Y	Y	Ν	Ν	Ν	
Observations	513,437	226,722	226,722	226,722	226,722	
R-squared	0.477	0.446	0.000	0.000	0.000	

This table estimates the effect of the change in one-year Treasury rate on most common types of time and saving deposits using Bauer and Swanson (2022) shocks as an instrument. The sample covers between 2000-2019.

Panel A: Persona	l Loans					
		Δ]	Loan Spread			
-	All \geq 2 Counties					
	(1)	(2)	(3)	(4)	(5)	
	0.100***	0 100***				
ΔR_t	0.179***	0.183***				
	(0.0222)	(0.0283)		0 (72**	0 770**	
$\Delta K_t \times HHI_c$	-0.291°	-0.683^{44}	-0.550^{m}	$-0.6/3^{m}$	-0.779^{44}	
	(0.154)	(0.251)	(0.271)	(0.302)	(0.317)	
Branch f.e.	Y	Y	Y	Y	Y	
County f.e.	Y	Y	Y	Y	Y	
Bank-time f.e.	Ν	Ν	Ν	Y	Y	
State-time f.e.	Ν	Ν	Y	Ν	Y	
Macro controls	Y	Y	Ν	Ν	Ν	
Observations	162,173	66,253	66,253	66,253	66,253	
R-squared	0.062	0.063	0.000	0.000	0.000	
Panel B: Automo	bile Loans					
		Δ	Loan Spread			
	All		\geq 2 Cou	nties		
	(1)	(2)	(3)	(4)	(5)	
		0.400444				
ΔR_t	0.245***	0.420***				
	(0.0222)	(0.0325)	0.005**	016		
$\Delta K_t \times HHI_c$	-0.189**	-0.341^{**}	-0.385**	-0.165	-0.277*	
	(0.0832)	(0.141)	(0.164)	(0.151)	(0.153)	
Branch f.e.	Y	Y	Y	Y	Y	
County f.e.	Y	Y	Y	Y	Y	
Bank-time f.e.	Ν	Ν	Ν	Y	Y	
State-time f.e.	Ν	Ν	Y	Ν	Y	
Macro controls	Y	Y	Ν	Ν	Ν	
Observations	76,695	34,030	34,030	34,030	34,030	
R-squared	0.059	0.077	0.001	0.000	0.000	

Table 5: Personal Loan and Automobile Loan Spreads

This table estimates the effect of the change in one-year Treasury rate on Personal and Automobile Loan spreads using Bauer and Swanson (2022) shocks as an instrument. The sample covers between 2000-2019.

Panel A: Home Equity Line of Credits (HELOC)					
		Δ	Loan Spread		
	All		\geq 2 Cou	nties	
	(1)	(2)	(3)	(4)	(5)
ΔR_t	0.969***	0.990***			
$\Delta R_t \times HHI_c$	(0.0131) -0.250** (0.0981)	(0.0179) -0.359** (0.149)	-0.357** (0.168)	-0.433** (0.191)	-0.513** (0.208)
Branch f.e. County f.e.	Y Y	Ŷ Ŷ	Y Y Y	Y Y Y	Y Y
Bank-time f.e. State-time f.e.	N N	N N	N Y	Y N	Y Y
Macro controls	Y	Y	Ν	Ν	Ν
Observations R-squared	143,330 0.208	64,526 0.175	64,526 0.000	64,526 0.000	64,526 0.000

Table 6: HELOC and Fixed Rate Mortgage Spreads

Panel B: 30-Year Fixed Rate Mortgages

		ΔΙ	.oan Spread		
-	All		\geq 2 Cour	nties	
-	(1)	(2)	(3)	(4)	(5)
ΔR_t	0.0249**	0.00325			
	(0.0102)	(0.0120)			
$\Delta R_t \times HHI_c$	0.00855	-0.101	-0.210*	-0.205*	-0.327***
	(0.0823)	(0.109)	(0.113)	(0.111)	(0.116)
Branch f.e.	Y	Y	Y	Y	Y
County f.e.	Y	Y	Y	Y	Y
Bank-time f.e.	Ν	Ν	Ν	Y	Y
State-time f.e.	Ν	Ν	Y	Ν	Y
Macro controls	Y	Y	Ν	Ν	Ν
Observations	39,554	23,195	23,195	23,195	23,195
R-squared	0.080	0.082	0.000	0.001	0.000

This table estimates the effect of the change in one-year Treasury rate on HELOCs and Fixed Rate Mortgage spreads using Bauer and Swanson (2022) shocks as an instrument. The sample covers between 2000-2019.

	∆Deposit Spread	∆WF Spread	∆Funding Spread	∆Loan Spread	ΔNIM	ΔROA
	(1)	(2)	(3)	(4)	(5)	(6)
ΔR_t	.880***	.707***	.867***	.278***	0297***	.00108
	(.00247)	(.0168)	(.00248)	(.00506)	(.00313)	(.00813)
$\Delta R_t \times BankHHI_{jt-1}$.0790***	.199*	.0996***	0767**	.0384**	.00961**
,	(.0150)	(.104)	(.0150)	(.0327)	(.0196)	(.00489)
Bank f.e.	Y	Y	Y	Y	Y	Y
Bank-level controls	Y	Y	Y	Y	Y	Y
Macro controls	Y	Y	Y	Y	Y	Y
Observations	455,505	455,505	455,505	455,505	455,505	455,505
R-squared	0.501	0.027	0.493	0.007	0.018	0.018
Panel B: Bank Assets	and Liabili	ities				
	∆Retail	ΔW .sale	∆Total	∆Total	∆Total	∆Total
	Deposits	Funding	Liabilities	Assets	Loans	Securities

Table 7: Bank-level Results

Panel A: Bank Interest Spreads

	Deposits	Funding	Liabilities	Assets	Loans	Securities
	(1)	(2)	(3)	(4)	(5)	(6)
ΔR_t	267***	.743*	408***	325***	188***	135
	(.0438)	(.0430)	(.0437)	(.0387)	(.0432)	(.128)
$\Delta R_t \times BankHHI_{it-1}$	0267	8.721***	.297	.109	.0325	2.719***
,	(.284)	(2.458)	(.282)	(.251)	(.304)	(.749)
Bank f.e.	Y	Y	Y	Y	Y	Y
Bank-level controls	Y	Y	Y	Y	Y	Y
Macro controls	Y	Y	Y	Y	Y	Y
Observations	455,505	455,505	455,505	455,505	455,505	455,505
R-squared	0.042	0.002	0.041	0.037	0.066	0.020

This table estimates the effect of the bank market power on bank-level outcomes. Panel A reports the results for the interest spreads and profitability. Panel B reports the results for assets and liabilities. ΔR_t is the change in the one-year Treasury Yield instrumented by Bauer and Swanson (2022) shocks. The sample covers between 2000-2019.

	(1)	(2)	(3)	(4)	(5)
	ΔLending	ΔLending	ΔLending	ΔLending	ΔLending
ΔR_t	-0.138***	-0.140***			
	(0.0134)	(0.0133)			
$\Delta R_t \times BankHHI_{it-1}$	0.290**	0.295**	0.273**	0.280**	0.292**
,	(0.118)	(0.120)	(0.125)	(0.128)	(0.128)
$\Delta R_t \times HHI_{ct-1}$		-0.0278		-0.0321	
		(0.0314)		(0.0315)	
County f.e	Y	Y	Y	Y	Y
Bank f.e	Y	Y	Y	Y	Y
County-bank f.e.	Y	Y	Y	Y	Y
County-time f.e.	Ν	Ν	Ν	Ν	Y
Time f.e.	Ν	Ν	Y	Y	Y
Macro-level controls	Y	Y	Ν	Ν	Ν
Observations	550,840	550,840	550,840	550,840	550,840
R-squared	0.031	0.031	0.003	0.003	0.003

Table 8: Table : Change in Bank-County Level Small Business Lending

This table estimates the effect of the bank market power on small business lending. Δ Lending is the percentage change in the total amount of small business lending originated by a given bank in a given county compared to the previous year. Bank-HHI is the bank's market power, and HHI is the concentration of the county where the bank branch operates. ΔR_t is the change in the one-year Treasury Yield instrumented by Bauer and Swanson (2022) shocks. The sample covers between 2000-2019.





The plots show the impulse responses of total county-level small business lending using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is four years. The period is 1997-2019. The control variables are changes in the national level of GDP, unemployment rate, inflation, and HHI_c . Standard errors are clustered by bank and county. HHI-0.15, HHI-0.20, and HHI-0.30 refer to counties with County-HHI indexes of 0.15, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of total county-level unemployment using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is four years. The period is 1997-2019. The control variables are changes in the national level of GDP, unemployment rate, inflation, and HHI_c . Standard errors are clustered by bank and county. HHI-0.15, HHI-0.20, and HHI-0.30 refer to counties with County-HHI indexes of 0.15, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.

Figure 9: Comparison of Model with Data-Change in the Loan Rate

This figure compares the model's prediction for the change in the loan and deposit rates with the ones obtained from the data. The model predictions are obtained by plugging the wholesale funding spread obtained from actual data into the model implied loan and deposit rate equations (See equations (19) and (20)). High-HHI refers to banks above the 75th percentile of the Bank-HHI distribution, whereas Low-HHI refers to banks below the 75th percentile of the Bank-HHI distribution. The average loan rate for the High-HHI group is calculated by averaging the loan rate of all banks above the 75th percentile of the Bank-HHI distribution for a given year. The data is from Call Reports spanning the period between 2000-2019.



(a) Change in the Loan Rate

(b) Change in the Deposit Rate



Appendix

A Data

A.1 Definition of the Bank-level Variables

This section describes my data construction procedure from the quarterly Call Reports. When constructing my sample. I control for bank mergers in my analysis by excluding banks with an asset growth rate of more than %100 between quarters.

Variable Name	Definition
Domestic Deposits	Saving Deposits+Time Deposits+Transaction Deposits
Wholesale Funding	Liabilities-Domestic Deposits
Deposit Rate	Interest Expense on Domestic Deposits/Domestic Deposits
Wholesale F. Rate	Interest Expense on Wholesale Funding/Wholesale Funding
Loan Rate	Interest Income on Total Loans/Total Loans
Deposit Spread	Federal Funds Rate-Deposit Rate
Loan Spread	Loan Rate-Treasury Yield with the Respective Maturity
Wholesale F. Spread	Federal Funds Rate-Wholesale Funding Rate
Liquidity	Cash+Securities+Federal Funds Repos
ROA	Net Income/Assets
NIM	Interest Rate on Assets-Interest Rate on Liabilities

Table A.1: Description of Banking Variables

A.2 Monetary Policy Shocks

I use high-frequency monetary policy shocks from Bauer and Swanson (2022) obtained through the first principal component analysis of Eurodollar futures contracts, ED1–ED4. I confirm the robustness of my results by using the orthogonalized version of these shocks with respect to economic news before the announcement. Figure A.1 plots these shocks over time.





This figure plots the Bauer and Swanson (2022) shocks over time.

B Robustness

This section provides results on various number of checks which confirm that the baseline results are robust to alternative estimation strategies, monetary policy shocks, usage of different samples, and market power measures. For the detailed discussion of the results see Section 7.

B.1 Results on Deposit and Loan Spreads

B.1.1 Using Alternative Products

Panel A: 06-Month CD									
	$\Delta Deposit Spread$								
	All		\geq 2 Cou	nties					
	(1)	(2)	(3)	(4)	(5)				
ΔR_t	0.380***	0.368*** (0.00521)							
$\Delta R_t \times HHI_c$	0.140*** (0.0295)	0.216*** (0.0399)	0.199*** (0.0367)	0.131*** (0.0403)	0.123*** (0.0399)				
Branch f.e.	Y	Y	Y	Y	Y				
County f.e.	Y	Y	Y	Y	Y				
Bank-time f.e.	Ν	Ν	Ν	Y	Y				
State-time f.e.	Ν	Ν	Y	Ν	Y				
Macro controls	Y	Y	Ν	Ν	Ν				
Observations R-squared	483,946 0.386	211,039 0.356	211,039 0.001	211,039 0.001	211,039 0.001				
Panel B: 10K Mo	ney Market F	unds							
		Δ]	Loan Spread						
	All		\geq 2 Cou	nties					
	(1)	(2)	(3)	(4)	(5)				

Table B.1: Time and Saving Deposit Spreads

	All \geq 2 Counties								
	(1)	(2)	(3)	(4)	(5)				
ΔR_t	0.890***	0.867***							
$\Delta R_t \times HHI_c$	$\begin{array}{c} (0.00399) \\ 0.214^{***} \\ (0.0289) \end{array}$	(0.00340) 0.193^{***} (0.0419)	0.260*** (0.0430)	0.227*** (0.0468)	0.229*** (0.0463)				
Branch f.e.	Y	Y	Y	Y	Y				
County f.e.	Y	Y	Y	Y	Y				
Bank-time f.e.	Ν	Ν	Ν	Y	Y				
State-time f.e.	Ν	Ν	Y	Ν	Y				
Macro controls	Y	Y	Ν	Ν	Ν				
Observations	512,650	226,212	226,212	226,212	226,212				
R-squared	0.634	0.600	0.001	0.000	0.000				

15-Year Fixed Rate Mortgages										
	Δ Loan Spread									
-	All		\geq 2 Cou	nties						
	(1)	(2)	(3)	(4)	(5)					
ΔR_t	0.140***	0.122***								
	(0.0119)	(0.0146)								
$\Delta R_t \times HHI_c$	-0.0371	-0.139	-0.245*	-0.236**	-0.321**					
	(0.0828)	(0.115)	(0.128)	(0.117)	(0.133)					
Branch f.e.	Y	Y	Y	Y	Y					
County f.e.	Y	Y	Y	Y	Y					
Bank-time f.e.	Ν	Ν	Ν	Y	Y					
State-time f.e.	Ν	Ν	Y	Ν	Y					
Macro controls	Y	Y	Ν	Ν	Ν					
Observations	34,352	18,923	18,923	18,923	18,923					
R-squared	0.093	0.071	0.001	0.001	0.001					

Table B.2: Loan Spreads

Tables B.1 and B.2 estimate the effect of the change in one-year Treasury rate on alternative deposit and loan products using Bauer and Swanson (2022) shocks as an instrument. The sample covers between 2000-2019.

B.1.2 Using Deposit Market Power

Panel B:Deposit Spread								
_	Δ Deposit Spread							
-	25K MMF	10K	MMF 12	2-Month CD	6-Month CD			
	(1)		(2)	(3)	(4)			
	0 250***	0.7	120***	0 105***	0 196***			
$\Delta K_t \times IIIII_c$	(0.0507)	(0.2	.0470)	(0.0383)	(0.0406)			
			Ň	Ň	Ň			
Branch f.e.	Ŷ		Ŷ	Y	Ŷ			
County f.e.	Y		Y	Y	Y			
Bank-time f.e.	Y		Y	Y	Y			
State-time f.e.	Y		Y	Y	Y			
Observations	226,722	22	26,212	205,920	211,039			
R-squared	0.000		0.000	0.000	0.001			
Panel A:Loan Sp	preads							
	Δ Loan Spread							
-	Personal	Auto	15-Yea	ar 30-Year	: 80-LTV			
	Loans	Loans	Mortgage	es Mortgages	B HELOCS			
	(1)	(2)	(3	3) (4)	(5)			
$\Delta R_t \times HHI_c$	-0.840***	-0.248*	-0.305*	-0.307***	-0.486**			
	(0.313)	(0.141)	(0.124	4) (0.112)	(0.195)			
Branch f.e.	Y	Y		Y Y	Ϋ́Υ			
County f.e.	Y	Y		Y Y	Y Y			
Bank-time f.e.	Y	Y		Y Y	Ý Y			
State-time f.e.	Y	Y		Y Y	Y Y			
Observations	66 253	34 030	18 97	23 23 195	64 526			
R-squared	0.000	0.000	0.00	01 0.000	0.000			

Table B.3: Loan and Deposit Spreads

This table estimates the effect of the change in one-year Treasury rate on loan and deposit spreads using Bauer and Swanson (2022) shocks as an instrument. HHI_c measures the market concentration of the county where the branch is located using deposit shares of the branches as a robustness exercise to Table 4 to 6. The sample covers between 2000-2019.

B.1.3 Using Alternative Policy Measures

Panel A: 1-Year T	Treasury Yield							
	$\Delta Deposit Spread$							
_	25K MMF	10K MMF	12-Month CD	6-Month CD				
	(1)	(2)	(3)	(4)				
$\Delta R_t \times HHI_c$	0.152*** (0.0265)	0.141*** (0.0250)	0.0892*** (0.0249)	0.128*** (0.0264)				
Branch f.e. County f.e. Bank-time f.e. State-time f.e.	Y Y Y Y	Y Y Y Y	Y Y Y Y	Y Y Y Y				
Observations R-squared	226,722 0.789	226,212 0.824	205,920 0.705	211,039 0.699				
Panel B: IV with	3: IV with orthogonalized shocks ∆Deposit Spread							

Table B.4: Deposit Spreads

	a Deposit Spread							
	25K MMF	10K MMF	12-Month CD	6-Month CD				
	(1)	(2)	(3)	(4)				
$\Delta R_t \times HHI_c$	0.363*** (0.0721)	0.348*** (0.0965)	0.145* (0.0883)	0.120 (0.0855)				
Branch f.e.	Y	Y	Y	Y				
County f.e.	Y	Y	Y	Y				
Bank-time f.e.	Y	Y	Y	Y				
State-time f.e.	Y	Y	Y	Y				
Observations R-squared	226,722 0.001	226,212 0.001	205,920 0.000	211,039 0.001				

This table estimates the effect of the change in the one-year Treasury rate and orthogonalized Bauer and Swanson (2022) shocks on deposit rates as a robustness exercise to Table 4. The sample covers between 2000-2019.

Panel A: 1-Year Treasury Yield									
	Δ Loan Spread								
	Personal Loans	PersonalAuto15-Year30-YearLoansLoansMortgagesMortgages							
	(1)	(2)	(3)	(4)	(5)				
$\Delta R_t \times HHI_c$	-0.330** (0.159)	-0.0539 (0.0678)	-0.169** (0.0706)	-0.231*** (0.0667)	-0.448*** (0.116)				
Branch f.e. County f.e. Bank-time f.e. State-time f.e.	Y Y Y Y	Y Y Y Y	Y Y Y Y	Y Y Y Y	Y Y Y Y				
Observations R-squared	66,253 0.408	34,030 0.594	18,923 0.630	23,195 0.563	64,526 0.486				

Table B.5: Loan Spreads

Panel B: IV with orthogonalized shocks

	∆Loan Spread							
	Personal	Auto	15-Year	30-Year	80-LTV			
	Loans	Loans	Mortgages	Mortgages	HELOCs			
	(1)	(2)	(3)	(4)	(5)			
$\Delta R_t \times HHI_c$	-1.692*	-0.277*	-0.466	0.437	-1.408*			
	(1.003)	(0.153)	(0.323)	(0.458)	(0.818)			
Branch f.e.	Y	Y	Y	Y	Y			
County f.e.	Y	Y	Y	Y	Y			
Bank-time f.e.	Y	Y	Y	Y	Y			
State-time f.e.	Y	Y	Y	Y	Y			
Observations	66,253	34,030	18,923	23,195	64,526			
R-squared	0.002	0.000	0.001	0.000	0.001			

This table estimates the effect of the change in the one-year Treasury rate and orthogonalized Bauer and Swanson (2022) shocks on loan rates as a robustness exercise to Tables 6 and 7. Δ spread is the change in branch-level loan spread, which is equal to the change in loan rate minus the respective Treasury yield that matches the loan's maturity. *HHI*_c measures the market concentration of the county where the branch is located. The sample covers between 2000-2019.

B.2 Results on Bank-level Outcomes

B.2.1 Using LP-IV strategy

Figure B.1: Impulse Responses of Bank-level Deposit Rates



The plots show the impulse responses of bank-level deposit rates using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level loan rates using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.



Figure B.3: Impulse Responses of Bank-level Wholesale Funding Rates

The plots show the impulse responses of bank-level wholesale funding rates using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.

Figure B.4: Impulse Responses of Bank-level NIMs



The plots show the impulse responses of bank-level net interest margins using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level assets using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level loans using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level securities using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level deposits using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.

Figure B.9: Impulse Responses of Bank-level Wholesale Funding



The plots show the impulse responses of bank-level wholesale funding using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.





The plots show the impulse responses of bank-level liabilities using local projection regressions with one-year Treasury yield instrumented by Bauer and Swanson (2022) shocks. The time horizon is eight quarters. The period is 2000-2019. The control variables are the change in the national level of GDP, unemployment rate, inflation, lagged change in assets, liquidity, and equity. Standard errors are clustered by bank. HHI-0.10, HHI-0.20, and HHI-0.30 refer to banks with Bank-HHI indexes of 0.10, 0.20, and 0.30, respectively. Shaded areas in panels (a) and (b) show 95% confidence intervals.

B.2.2 Using Additional Bank-level Controls

Panel A: Bank Interest Spreads

	∆Deposit Spread	∆WF Spread	∆Funding Spread	∆Loan Spread	ΔNIM	ΔROA
	(1)	(2)	(3)	(4)	(5)	(6)
ΔR_t	.838***	.676***	.825**	.206***	0273***	.00151*
	(.00247)	(.0162)	(.00248)	(.00495)	(.00309)	(.000795)
$\Delta R_t \times BankHHI_{jt-1}$.0723***	.209*	.0775***	0658*	.0694***	.0138***
,	(.0170)	(.108)	(.0157)	(.0336)	(.0199)	(.00504)
Bank f.e.	Y	Y	Y	Y	Y	Y
Bank-level controls	Y	Y	Y	Y	Y	Y
Macro controls	Y	Y	Y	Y	Y	Y
Observations	455,505	455,505	455,505	455,505	455,505	455,505
R-squared	0.506	0.024	0.542	0.134	0.006	0.002

Table B.6: Bank-level Results

Panel B: Bank Assets and Liabilities

	∆Retail Deposits	∆W.sale Funding	∆Total Liabilities	∆Total Assets	∆Total Loans	∆Total Securities
	(1)	(2)	(3)	(4)	(5)	(6)
ΔR_t	613***	.0432	614***	514***	370***	218*
	(.0453)	(.413)	(.0445)	(.0392)	(.0443)	(.127)
$\Delta R_t \times BankHHI_{jt-1}$.476	6.410**	.584*	.375	.0645	3.308***
,	(.307)	(2.745)	(.307)	(.272)	(.329)	(.791)
Bank f.e.	Y	Y	Y	Y	Y	Y
Bank-level controls	Y	Y	Y	Y	Y	Y
Macro controls	Y	Y	Y	Y	Y	Y
Observations	455,505	455,505	455,505	455,505	455 <i>,</i> 505	455,505
R-squared	0.007	0.004	0.004	0.004	0.007	0.005

This table estimates the effect of the bank market power on bank-level outcomes. Panel A reports the results for the interest spreads and profitability. Panel B reports the results for assets and liabilities. ΔR_t is the change in the one-year Treasury Yield instrumented by Bauer and Swanson (2022) shocks. The sample covers between 2000-2019.

B.3 Using Time Fixed Effect Specification

Panel A: Bank Interest Spreads

	∆Deposit Spread	∆WF Spread	∆Funding Spread	∆Loan Spread	ΔNIM	ΔROA		
	(1)	(2)	(3)	(4)	(5)	(6)		
$\Delta R_t \times BankHHI_{jt-1}$.0570***	.149*	.0789***	104***	.0560***	.0132***		
,	(.0143)	(.0882)	(.0144)	(.0323)	(.0178)	(.00460)		
Bank f.e.	Y	Y	Y	Y	Y	Y		
Bank-level controls	Y	Y	Y	Y	Y	Y		
Macro controls	Y	Y	Y	Y	Y	Y		
Observations	455,505	455,505	455,505	455,505	455,505	455,505		
R-squared	0.051	0.006	0.063	0.037	0.015	0.019		
Panel B: Bank Assets	Panel B: Bank Assets and Liabilities							
		A T A 7 1	ATT (1	ATT (1	ATT (1	ATT 1 1		

Table B.7: Bank-level Results

	∆Retail Deposits	∆W.sale Funding	∆Total Liabilities	∆Total Assets	∆Total Loans	∆Total Securities
	(1)	(2)	(3)	(4)	(5)	(6)
$\Delta R_t \times BankHHI_{it-1}$	0127	8.436***	.310	.114	.0756	1.869**
,	(.283)	(2.461)	(.282)	(.303)	(.251)	(.745)
Bank f.e.	Y	Y	Y	Y	Y	Y
Bank-level controls	Y	Y	Y	Y	Y	Y
Macro controls	Y	Y	Y	Y	Y	Y
Observations	455,505	455,505	455,505	455,505	455,505	455,505
R-squared	0.036	0.001	0.036	0.045	0.042	0.019

This table estimates the effect of the bank market power on bank-level outcomes. Panel A reports the results for the interest spreads and profitability. Panel B reports the results for assets and liabilities. ΔR_t is the change in the one-year Treasury Yield instrumented by Bauer and Swanson (2022) shocks. The sample covers between 2000-2019.

B.4 Results on Bank-County Level Small Business Lending

B.5 Using the Full Sample

This section provides results on the impact of the bank market power on new small business lending using the full sample that includes loan originations less than \$100,000 in value.

	(1)	(2)	(3)	(4)	(5)
	∆Lending	ΔLending	ΔLending	ΔLending	∆Lending
ΔR_t	-0.158***	-0.160***			
	(0.0438)	(0.0418)			
$\Delta R_t \times BankHHI_{it-1}$	0.486**	0.490**	0.468**	0.472**	0.513**
,	(0.197)	(0.198)	(0.195)	(0.195)	(0.202)
$\Delta R_t \times HHI_{ct-1}$		-0.0230		-0.0228	
		(0.0412)		(0.0374)	
County f.e	Y	Y	Y	Y	Y
Bank f.e	Y	Y	Y	Y	Y
County-bank f.e.	Y	Y	Y	Y	Y
County-time f.e.	Ν	Ν	Ν	Ν	Y
Time f.e.	Ν	Ν	Y	Y	Y
Macro-level controls	Y	Y	Ν	Ν	Ν
Observations	921,233	921,233	921,233	921,233	921,233
R-squared	0.022	0.022	0.002	0.002	0.002

Table B.8: Table Change in the Bank-County Level Small Business Lending

This table estimates the effect of the bank market power on new small business lending using the full sample that includes loan originations less than \$100,000 in value. Δ Lending is the percent of the change in the total amount of small business lending originated by a given bank in a given county compared to the previous year. ΔR_t is the change in the one-year Treasury Yield instrumented by Bauer and Swanson (2022) shocks. Bank-HHI is the bank's market power, and HHI is the concentration of the county where the bank operates. The sample covers between 2000-2019.

C Model

In this section, I derived the CES demand functions used in 6.1

C.1 Dixit-Stiglitz Aggregator:

The solution to loan and deposit demand comes from the Dixit-Stiglitz aggregator illustrated below:

Loan Market:

Borrower seeks a total amount of loans equal to L; he borrows an amount L_j from each bank j and faces the following constraint:

$$L_{j} = \left(\frac{1}{N}\sum_{j=1}^{N}L_{j}\frac{\theta^{\ell}-1}{\theta^{\ell}}\right)^{\frac{\theta^{\ell}}{\theta^{\ell}-1}}$$
(22)

Where $1 < \theta^{\ell}$ is the elasticity of substitution between banks.

Demand for the borrower can be derived from minimizing over L_j the total repayment (including principal) due to a continuum of banks *j*:

$$\min_{L_j} \frac{1}{N} \sum_{j=1}^N (1+i_j^L) L_j$$

subject to

$$\left(\frac{1}{N}\sum_{j=1}^{N}L_{j}^{\frac{\theta^{\ell}-1}{\theta^{\ell}}}\right)^{\frac{\theta^{\ell}}{\theta^{\ell}-1}} \geq L$$

FOC with respect to L_i yields loan demand:

$$L_j = \left(\frac{1+i_j^L}{1+i^L}\right)^{-\theta^\ell} L$$

Where

$$1 + i^{L} = \left(\frac{1}{N}\sum_{j=1}^{N}i_{j}^{L^{1-\theta^{\ell}}}\right)^{\frac{1}{1-\theta^{\ell}}}$$

Deposit Market:

Savers want to maximize total repayment from deposits subject to total deposits as aggregated through a CES aggregator:

$$D_{j} = \left(\frac{1}{N}\sum_{j=1}^{N}D_{j}\frac{\theta^{d}-1}{\theta^{d}}\right)^{\frac{\theta^{d}}{\theta^{d}-1}}$$
(23)

Where $\theta^d < -1$ is the elasticity of deposit substitution across banks.

$$\max_{D_j} \quad \frac{1}{N} \sum_{j=1}^N (1+i_j^D) D_j$$

subject to

$$\left(\frac{1}{N}\sum_{j=1}^{N}D_{j}^{\frac{\theta^{d}-1}{\theta^{d}}}\right)^{\frac{\theta^{d}}{\theta^{d}-1}} \leq D$$

FOC with respect to D_j yields deposit supply:

$$D_j = \left(\frac{1+i_j^D}{1+i^D}\right)^{-\theta^d} D$$

Where

$$1+i^D = \left(\frac{1}{N}\sum_{j=1}^N i_j^{D^{1-\theta^d}}\right)^{\frac{1}{1-\theta^d}}$$